Volume 54 Number 10 21 May 2025 Pages 4619-5148

Chem Soc Rev

Chemical Society Reviews

rsc.li/chem-soc-rev



ISSN 0306-0012



REVIEW ARTICLE Mehrdad Asgari, David Fairen-Jimenez *et al.* The structuring of porous reticular materials for energy applications at industrial scales

Chem Soc Rev



View Article Online

REVIEW ARTICLE

Check for updates

Cite this: Chem. Soc. Rev., 2025, 54, 4701

The structuring of porous reticular materials for energy applications at industrial scales

Mehrdad Asgari, ()*** Pablo Albacete, ()*** Dhruv Menon, ()*** Yuexi Lyu,** Xu Chen ()**** And David Fairen-Jimenez ()***

Reticular synthesis constructs crystalline architectures by linking molecular building blocks with robust bonds. This process gave rise to reticular chemistry and permanently porous solids. Such precise control over pore shape, size and surface chemistry makes reticular materials versatile for gas storage, separation, catalysis, sensing, and healthcare applications. Despite their potential, the transition from laboratory to industrial applications remains largely limited. Among various factors contributing to this translational gap, the challenges associated with their formulation through structuring and densification for industrial compatibility are significant yet underexplored areas. Here, we focus on the shaping strategies for porous reticular materials, particularly metal-organic frameworks (MOFs) and covalent organic frameworks (COFs), to facilitate their industrial application. We explore techniques that preserve functionality and ensure durability under rigorous industrial conditions. The discussion highlights various configurations - granules, monoliths, pellets, thin films, gels, foams, and glasses - structured to maintain the materials' intrinsic microscopic properties at a macroscopic level. We examine the foundational theory and principles behind these shapes and structures, employing both in situ and post-synthetic methods. Through case studies, we demonstrate the performance of these materials in real-world settings, offering a structuring blueprint to inform the selection of techniques and shapes for diverse applications. Ultimately, we argue that advancing structuring strategies for porous reticular materials is key to closing the gap between laboratory research and industrial utilization

Received 12th February 2025 DOI: 10.1039/d5cs00166h

rsc.li/chem-soc-rev

^a The Adsorption and Advanced Materials Laboratory (A²ML), Department of Chemical Engineering and Biotechnology, University of Cambridge, Philippa Fawcett Drive, Cambridge, CB3 0AS, UK. E-mail: ma2000@cam.ac.uk, df334@cam.ac.uk

^b Departamento de Química Inorgánica, Universidad Autónoma de Madrid, 28049 Madrid, Spain



Mehrdad Asgari

Mehrdad Asgari Dr is а postdoctoral researcher working under Prof. David Fairén-Jiménez in the Adsorption & Advanced Materials Laboratory (A^2ML) at the University of Cambridge. He specializes in developing monolithic MOFs for chemical separation and catalysis. Dr Asgari earned his PhD from EPFL in 2020 under Prof. Wendy Queen, focusing on carbon capture, and continued as a postdoctoral researcher with

Profs. Berend Smit and François Maréchal, applying computational methods to investigate gas adsorption phenomena. Awarded both an SNSF Early Postdoctoral Mobility Fellowship and a Marie Skłodowska-Curie Fellowship, his research bridges experimental synthesis with computational simulations to advance sustainable MOF-based technologies.



Dr Pablo Albacete completed his PhD at the Universidad Autónoma de Madrid (2020) under Prof. Félix Zamora and Dr Ana Platero-Prats, exploring structure-property relationships in 2D covalent organic frameworks (COFs). He later joined Prof. David Fairén-Jiménez at the A²ML, University of Cambridge, as a Margarita Salas Fellow, where he developed expertise in synthesizing and shaping monolithic MOFs, COFs, and MOF-COF composites. In 2024, he returned to Madrid to

Pablo Albacete

continue his research. Passionate about science communication, he actively engages in outreach projects to make science accessible and understandable to broader audiences.

1. Introduction

Establishing precise control over the size and uniformity of the porous space has multiple implications for large-scale applications. Materials that exhibit a high degree of porosity are of interest across diverse technologies due to their ability to interact with other chemical species, such as gases and liquids, not only at their external surface but also throughout their internal porosity.¹ Beyond classical porous materials such as activated carbons and zeolites, the synthesis of permanently porous structures in reticular materials or coordination polymers was, until the late 1990s, thought to be largely unfeasible based on the age-old perception that 'nature abhors vacuum'.² The emergence of the 'reticular chemistry' concept facilitated the geometry-guided design of periodically extended, crystal-line structures by linking molecular building blocks through

journey that lies ahead, he aspires to bridge the translational gap

between laboratory research and industrial applications.

strong bonds – leading to the creation of new porous materials. These rigid building blocks assemble into predetermined target networks, retaining their structural integrity and rigidity throughout the synthesis process.^{3–5} The presence of strong bonds facilitates the generation of crystalline frameworks with high architectural stability – overcoming a critical challenge that previously held back the realization of permanently porous solids.⁶ This pioneering class of reticular materials were referred to as metal–organic frameworks (MOFs)⁷ or porous coordination polymers (PCPs),⁸ but also expanded into sister families such as covalent organic frameworks (COFs)⁹ and metal–organic polyhedra (MOPs)¹⁰ – summarized in Fig. 1 – as well as porous organic cages (POCs) and hydrogen-bonded organic frameworks (HOFs).

MOFs and PCPs are constructed by connecting metal ions or metal-containing nodes with organic linker molecules using



Dhruv Menon

Dhruv Menon is a doctoral candidate in Chemical Engineering and Biotechnology at the A^2ML , University of Cambridge, supervised by Prof. David Fairén-Jiménez. He previously earned a master's degree in physical Cavendish sciences from the Cambridge. His Laboratory at research interests integrate computational approaches for molecular design and engineering, spanning biology, chemistry and energy-related applications. In the

Yuexi Lyu

Yuexi Lyu is a doctoral candidate in Chemical Engineering and Biotechnology at the A^2ML , University of Cambridge, supervised by Prof. David Fairén-Jiménez. Her research centers on synthesizing, optimizing, and characterizing nanoscale MOFs, with a primary interest in biomedical applications such as drug delivery and photodynamic therapy (PDT). She aims to improve the stability, biocompatibility, and therapeutic efficacy MOF-based of nanomaterials,

contributing significantly to the fields of nanomedicine and targeted cancer therapies.



Xu Chen

Dr Xu Chen obtained his PhD from Shanghai Jiao Tong University (2018) under Prof. Yong Cui, focusing on MOFs and catalysis. He later joined Prof. David Fairén-Jiménez at the A^2ML , University of Cambridge. His current research interests include materials discovery for applications in drug delivery, carbon capture and hydrogen storage, leveraging his expertise to advance novel solutions in healthcare and energy sectors.



David Fairen-Jimenez

David Fairén-Jiménez is Professor of Molecular Engineering at the University of Cambridge, leading the Adsorption & Advanced Materials Laboratory (A^2ML). His computational research spans simulation, synthesis, and molecular biology for materials discovery, focusing on biomedical and applications monolithic materials for energy transition. After earning his PhD in sol-gel, monolithic carbon aerogels from the University of Granada, he con-

ducted postdoctoral research at the University of Edinburgh and Northwestern University. He is committed to translating research into industry; he has founded two spin-out companies, Immaterial, specializing in carbon capture and hydrogen storage, and Vector Bioscience, focused on advancing precision drug delivery solutions.





Fig. 1 An introduction to porous reticular materials. (a) MOFs or PCPs are metal ions or metal-containing nodes connected by organic linker molecules using coordination bonds. Due to the large length of the linker molecules, MOFs show a high degree of porosity. Here, we present crystal structures of archetypal MOFs with high porosities: PCN-222, UiO-66, and MOF-808. (b) COFs are covalent porous crystalline polymers formed by the integration of organic building blocks into ordered structures. These structures are typically lightweight and tend to have low mass densities. Here, we present crystal structures of archetypal COFs: COF-42 and COF-300. (c) MOPs are coordination cage compounds formed through linking metal cations with organic linkers. As opposed to MOFs, these cages exist in isolation – however, they can serve as building blocks for creating extended solids. (d) Outside of these classes of reticular materials, in recent years, there has been developments into new families such as POCs and HOFs. Figure has been adapted with permission from ref. 11 Copyright 2015 American Chemical Society, ref. 12 Copyright 2015 Oxford Academic, ref. 10 Copyright 2021 Royal Society of Chemistry, and ref. 13 Copyright Elsevier 2022.

coordination bonds (Fig. 1a). Due to the relatively strong bonds and potentially large length of the linker molecules, MOFs can show a high degree of porosity (up to 90% of void volume), and large specific surface areas (1000–10 000 m² g⁻¹).⁷ In addition, the high flexibility in terms of the choice of precursors facilitates the realization of virtually infinite possible structures, each tailored to the application at hand. So far, more than 100 000 PCPs/MOFs have been reported in the literature, and about 15% of them are porous.^{7,14} Similar principles of topology-guided design have led to the synthesis of COFs, covalent-bonded, porous crystalline polymers that integrated organic building blocks into ordered structures (Fig. 1b).⁹ The

use of covalent bonds to connect molecular building blocks has, historically, led to the construction of amorphous or poorly defined materials. However, in the case of COFs, this problem was overcome by the use of B-O, C-N, B-N, and B-O-Si linkages.¹¹ In addition, MOPs are coordination cage compounds formed by the linkage of metal cations and organic linkers that, in contrast to MOFs, exist as isolated cages. MOPs can also be used as secondary building blocks to create extended solids, such as MOFs (Fig. 1c).^{10,15} Like MOPs, POCs are discrete, cage-like structures created through the covalent bonding of organic molecules, without metals. These structures are notable for their solution processability and adjustable pore sizes. On the other hand, HOFs represent a class of porous materials formed by hydrogen bonding between organic molecules. Unlike the coordination bonds in MOFs or the covalent bonds in COFs, hydrogen bonds in HOFs provide a balance between structural rigidity and flexibility. This characteristic makes HOFs potentially valuable for applications requiring reversible assembly and disassembly, such as in molecular storage, and selective separation processes.¹³

While these new porous materials have potential for multiple applications, the underlying principles surrounding their selection are centered around their adsorption properties, In particular, adsorption properties for either the storage, separation (or capture), and release of chemical entities such as gases, ions, or liquids within their void space. Fig. 2 shows key, potential, energy-related applications for reticular materials that are covered in this review: (i) gas storage, (ii) gas separation, (iii) separation by membranes, (iv) heterogeneous catalysis, and (v) thin-film-based sensing. Gas storage is crucial in various industries, primarily concerning energy stability and alternative fuel sources.¹⁶ Nowadays, the most important gaseous energy vector of interest is, arguably, hydrogen. It is a key element in the necessary, green energy transition and decarbonization.¹⁷ Here, the high surface area and pore volume of porous materials are important, as they are strongly correlated to their gravimetric adsorption capacity. However, the material's density is another - very often ignored - parameter, as together with the pore volume, it defines the volumetric adsorption capacity: this is, the amount of gas one can store in a fixed amount of volume.¹⁶ Indeed, most industrial applications, including hydrogen storage and carbon capture, have limitations on the space that can be used to be implemented. Recent advancements in shaped, densified, MOFs (Fig. 2a) called monoliths due to their single-form factor - have surpassed methane storage targets set by the US Department of Energy, highlighting their potential to meet evolving energy demands while accommodating diverse functional requirements.¹⁸

For thin films, porous reticular materials have been proposed for sensing, electronics, and optics, enabling precise control and manipulation of light, electrons, and chemical species – impacting environmental monitoring, safety, and automation.²³ They have transformed these fields, as seen in 3D-MOF thin films (Fig. 2b), exhibiting high sensitivity and rapid response in ammonia sensing. While we discuss these

applications in some detail in the sections that follow, for better context, we refer readers to dedicated reviews on these topics.17,23-26 Porous reticular materials have also been proposed for gas separations such as carbon dioxide, carbon monoxide, ammonia, and hydrocarbons. Their porosity can be tuned to facilitate molecular sieving effects or promote specific interactions for selective adsorption.²⁴ For example, MOFs such as SIFSIX and NbOFIVE, constructed from fluorinated clusters and pyrazine/pyridine-based ligands, show their efficacy in hydrocarbon separation and carbon capture (Fig. 2c).²⁰ Their adjustable porosity enables precise molecular sieving effects and selective adsorption tailored to different gases. Moreover, the separation properties of reticular materials can also be incorporated into membranes, enabling molecular sieving of gases, ions (e.g., toxic, heavy metals), and liquids based on differences in size, shape, and chemical affinity.^{24,25} Fig. 2d shows a schematic diagram of MOF-based membranes incorporated into a polymer matrix, suitable for various applications including water purification. For heterogeneous catalysis, reticular materials can outperform conventional materials in certain reactions. Here, their potential is based on their high surface areas that allow high reaction rates, the presence of functional groups and metals to promote the reactions, and the tailorability of the porous structure to allow for rapid transport of reactant and product molecules while offering the possibility of shape-selective catalysis.²⁶ Techniques such as pyrolysis of MOF-based materials can enhance catalytic activity further by generating a porous carbon matrix containing single active catalytic sites (Fig. 2e).22

Despite the excellent performance metrics and promise of reticular materials, their successful translation to industry is limited. Following 25 years since their inception and very few notable exceptions,²⁷ their study is still largely confined to laboratory-based studies.²⁸ Indeed, the commercialization of materials technologies spans long timescales - with reports suggesting approximately 5-15 years to transition from discovery to the commercial market.²⁹ Technology Readiness Levels (TRLs) provide a strong framework for appreciating the associated costs and timescales for commercialization.²⁷ TRLs of 1-3 correspond to fundamental research, 4-6 imply applied research - focusing on prototyping and development, while TRLs of 7-9 focus on commercial-scale deployment (Fig. 3a).²⁷ Here, arguably, TRL 4-6 is the most challenging phase - often termed the 'valley of death' - as it lies between stages where public funding is limited while private capital is difficult to secure. More advanced MOF-based technologies - such as Svante's development of CALF-20 laminates³⁰ and Immaterial's sol-gel, monolithic solutions³¹ for CO₂ capture from wet acidic gas streams – lie in this phase.^{27,32} As such, we term the region between TRLs 1-3 and TRLs 7-9 as the 'translational gap'. Several aspects are responsible for this translational gap, including cost, hydrochemical, and mechanical stability. While cost is a function of raw materials, process synthesis and scale, stability is still one of the most important concerns for longterm operability - particularly in applications subject to harsh industrial conditions such as carbon capture.33 Crystalline





Fig. 2 Energy applications of porous reticular materials. (a) Gas storage: high surface areas and the prospect of incorporating multiple functionalities make reticular materials ideal for gas storage. In recent years, monolithic MOFs have made tremendous strides on this front, surpassing the target set by the US Department of Energy (dashed red line) for methane (CH₄) storage. (b) Thin-film sensing: porous reticular materials have been used in sensing, electronics, and optics, enabling precise control and manipulation of light, electrons, and chemical species. As an illustrative example, we present a 3D-MOF thin film with good crystallinity and precisely controllable thickness with high sensitivity and fast response for ammonia (NH₃) sensing. Adapted with permission from ref. 19 Copyright 2021 Wiley. (c) Gas separation: the porous nature of reticular materials makes them effective for complex gas separations. Here, we show the design of two MOFs – each specialized for a specific gas separation. SIFSIX effectively sieves acetylene over ethylene, while NbOFFIVE selectively sieves propylene over propane. Adapted with permission from ref. 20 Copyright 2016 American Association for the Advancement of Science. (d) Separation by membranes: filtration separation across a MOF-based membrane. These separations typically rely on the pore size or the rate of diffusion. Species smaller than the pore diameter permeate the membrane, while larger species are unable to permeate – leading to a selective separation. Adapted with permission from ref. 21 Copyright 2016 Springer Nature. (e) Heterogeneous catalysis: high surface areas allow high reaction rates per unit volume, and the tailorability of the porous structure allows rapid transport of reactant and product molecules. Here, we present the design of a MOF-based cobalt nanoparticle that catalyses the synthesis of amines. Adapted with permission from ref. 22 Copyright 2017 American Association for the Advancement of Science.

porous materials are considered metastable with respect to their dense phase.³³ In the case of MOFs, this refers to dense, amorphous phases obtained either through the thermal route³⁴

or *via* mechanical pressure.^{34,35} In this regard, the structuring and densification is an issue that – as stated above – has been relatively ignored in the academic literature but is critical in



Fig. 3 Translational gap towards the commercialisation of porous reticular materials. (a) Typical investments required at each TRL – with the slump at TRLs 4–6 corresponding to the translational gap. Adapted with permission from ref. 27 Copyright 2024 Springer Nature. Structuring as a crucial aspect of the translational gap. (b) A poorly chosen structuring technique – for instance mechanical pressure – may cause a collapse of the porosity. (c) Low densification may lead to interstitial voids.

industrial applications. Owing to this metastability, structuring is a crucial aspect because poorly chosen structuring techniques using, *e.g.*, mechanical pressure, often lead to a partial collapse in the material's porosity (Fig. 3b). On the other hand, poor structuring and densification can also lead to low densities due to the existence of interstitial space between the material particles – a volume that, generally, cannot be exploited in adsorption, increasing the footprint of designed systems, and reducing the promised volumetric performance of MOFs (Fig. 3c). Both options – pore collapse and low densification – are detrimental to their performance, ultimately defeating the purpose of employing porous materials (Fig. 3b and c).

The structuring and densification of reticular materials have broad implications, for their industrial application. In this review, we use the terms 'structuring' and 'shaping' interchangeably, as these processes are inherently correlated in the present context. While 'shaping' typically refers to the macroscopic transformation of materials into functional forms, 'structuring' focuses on maintaining and optimizing the internal architecture, including pore size, connectivity, and hierarchical organization. Unlike conventional materials, where shaping may compromise structural integrity, reticular materials must be shaped while preserving microstructural features to retain their properties at macro-scales. Traditional shaping techniques often risk pore collapse, reduced surface area, or loss of crystallinity, making it essential to incorporate structuring principles to maintain performance. Advanced processing techniques such as gel casting, additive manufacturing, and templated assembly simultaneously shape and structure these materials, ensuring their functional properties are retained. Given the close relationship between these concepts, and the fact that porous materials research often overlaps these definitions, we have adopted a flexible approach in using these terms throughout this review.

In this review, we explore various strategies for structuring porous reticular materials, including MOFs, COFs, HOFs, and other emerging frameworks, focusing on their implications for energy applications. It is important to acknowledge, however, that MOFs have received significantly more attention in both academic and industrial contexts. This is due to their earlier discovery, structural diversity, and well-established synthetic protocols, which have enabled their widespread adoption across applications of interest. Compared to other classes of reticular materials, MOFs have undergone extensive optimization and scaleup efforts, being, currently, the most advanced candidates for industrial implementation.²⁷ Consequently, many structuring techniques have been developed and refined specifically for MOFs, leading to their predominant focus in this review. However, we also incorporate key examples of COF structuring strategies to highlight the broader relevance of these methods across different porous frameworks. Our discussion aims to provide a strong understanding of structuring challenges and opportunities that are in principle, applicable across reticular materials, while placing a bulk of the emphasis on MOFs due to their closer proximity to large-scale applications.

We focus here on energy applications, but the underlying principles are universal, and the described techniques are transferable to applications and materials in different contexts. Section 2 focuses on the hierarchical synthesis of porous, reticular materials at the microscale and their size- and shape-controlled synthesis, templated approaches, and selfassembly of superstructures. Section 3 describes the landscape of available shapes for structuring porous reticular materials. Section 4 describes strategies for achieving these shapes at the macroscale and how these strategies would impact the material's performance. Section 5 discusses key considerations for the industrial translation. Here, we distinguished between the shape and the shaping technique because each form may be achieved using different techniques. At the end of the day, the choice of technique will depend on the scale of production, the associated costs, and the desired quality of the material at the macroscale. Considering the vast application landscape of these materials, rather than providing comprehensive discussions about specific applications, the goal is to provide a holistic, structuring 'blueprint' for bridging the translational gap.

2. The hierarchical synthesis of reticular materials at the microscale

A key feature of reticular materials is their metastable crystalline structure with empty pores. For industrial applications,

one needs to ensure that the energy barrier is high enough for the transition from a porous crystalline to an amorphous phase to avoid pore collapse,³³ or that there are alternative methods for shaping and densification that avoid this issue. In the first case, this requires the materials to survive high mechanical pressures, which necessitates control over the micro and submicroscale architecture of reticular materials.36 Given that these structures arise from coordination-driven molecular self-assembly,³⁷ the main challenge involves directing such reactions to create superstructures.³⁸ Advances in synthetic and fabrication technologies allow for significant control at these scales. Here, we introduce these techniques, setting the stage for discussing the structuring of reticular materials at the macroscopic level without covering the basic theory behind material formation. For in-depth discussions, we refer readers to specialized reviews on this topic.39,40

2.1. Size-controlled synthesis

In industrial applications, the performance of the macroscale, shaped material depends on the particle size - or rather the dispersity of the particle size - at the microscale. Good control over the dispersity could help reduce the interstitial space formed upon densification through a particle size distribution approach.^{41,42} However, while these processes are conceptually well-defined, conventional models may struggle to fully capture the complexities of nucleation and growth phenomena, especially in the case of MOFs and similar materials.43 Take, for example, the LaMer model,^{40,44} which distinguishes nucleation and growth, attributing distinct thermodynamic driving forces to each due to high precursor concentrations. While this model aligns with observations for certain MOFs, such as MOF-5,⁴⁵ where crystal nucleation and growth distinctly occur, for other MOFs, such as HKUST-1,⁴⁶ these phases appear to overlap, suggesting a simultaneous occurrence. Given these observations, Brozek et al.43 show that the kinetic control of chemical parameters to arrest particle growth is necessary for the modulation of particle size. The kinetic entrapment of MOF crystal size regimes relies on the interplay of competitive chemical equilibria, which encompass: (i) linker deprotonation, (ii) modulator deprotonation, (iii) linker complexation, and (iv) termination.⁴³ To form a metal-linker bond, typically, the linker must undergo deprotonation, and a similar deprotonation process is required for the modulator to function effectively. Furthermore, both the linker and the modulator compete for coordination sites with metal ions, and so, control over these two reactions allows regulation of particle size. The degree of complexation between the metal and the linker also plays a pivotal role in governing growth. The effective arrest of particle growth occurs when the concentration of the linker significantly surpasses that of the metal ion. This phenomenon arises from the fact that higher equivalent linker concentrations enhance metal-linker complexation, consequently depleting available metal-ion concentrations. In essence, higher concentrations of linkers or modulators tend to typically result in smaller nanocrystal sizes. In each of the four equilibria discussed above, the prevalence of fast-forward reaction rates and

limited reversibility tends to favor the formation of bulk structures. Modulators, compounds that can control the crystallization process are also important. Fig. 4a illustrates the impact of these parameters, showing the variation in crystal sizes of ZIF-8 across different concentrations of zinc metal, linker (1-methylimidazolium, Hmim), modulator (n-butylamine, n-BuNH₂).⁴³ Notably, excessive linker concentrations alone can diminish local metal ion availability, resulting in smaller crystal sizes even without a modulator present. This highlights the critical role of precursor ratios in finely modulating crystal growth dynamics in MOFs. Fig. 4a (right), shows a 'seesaw' relationship between the acidic linker or modulator concentrations and crystal size. Below a specific concentration threshold, increasing the ligand concentration reduces particle size due to metal ion depletion, showing the sensitivity of crystal growth to precursor ratios and local chemical environments. Conversely, surpassing this threshold can disrupt the deprotonation mechanism essential for crystal growth, leading to larger particle sizes. This dual effect emphasizes the rather delicate balance required in precursor concentrations to achieve desired crystal sizes in frameworks like ZIF-8. Such size control strategies could be applied to other porous reticular materials as well. For instance, the downsizing of PCN-224 from millimeter to nanoscale (Fig. 4b) dimensions through system dilution preserves phase purity and stoichiometry while promoting the formation of smaller MOF monomers.⁴⁷ Expanding to COF synthesis, the manipulation of modulator concentration is shown to facilitate size control of a spherical COF (Fig. 4c), thus representing another approach to tailor the physical characteristics of porous materials, offering insights into how chemical manipulation can influence their structural properties.48 In some cases, modulators work by altering the pH, thereby influencing the kinetics of linker deprotonation. Higher pH values accelerate linker deprotonation, leading to the formation of smaller particles. On the other hand, lower pH values slow down deprotonation, causing nucleation to occur over extended periods, and ultimately yielding larger crystals. In other cases, modulators act as coordination agents, competing with the linker during complexation.49

2.2. Shape and morphology-controlled synthesis

In addition to size control, modulators and templates modify the nucleation and growth mechanisms in porous materials, allowing the ability to control their shape and overall morphology.⁵⁰ Similar to size control, the main parameters that control shape are:

(i) Coordination modulation: this method involves competition between the modulator and the linker for metal ions. For controlling particle shape, higher modulator concentrations can slow down the precipitation rate of amorphous phases, thus improving crystallinity (Fig. 5a).^{49–51} Additionally, modulators physically prevent crystal aggregation, leading to anisotropic growth. This approach provides control over the morphology of the resulting crystals, enabling the formation of smaller, relatively uniform nanoparticles (NPs) in various shapes.⁵² Typically, additives with the same chemical functionality as the linker, such as



Fig. 4 Size-controlled synthesis of reticular materials. (a) Left: Trends in the crystal sizes of ZIF-8 upon varying concentrations of the metal (Zn), linker (1-methylimidazolium, Hmim), and the modulator (*n*-butylamine, *n*-BuNH₂). Excess concentrations of the linker deplete the local concentration of metal ions, thereby causing smaller crystal sizes even in the absence of a modulator. Right: There exists a 'seesaw' relationship between the crystal size and the concentration of acidic linker/modulator. At concentrations below the minima, increasing the ligand concentration leads to a decrease in particle size due to the depletion of local metal concentrations. At concentrations above the minima, increasing the ligand concentration interferes with the deprotonation mechanism. Adapted (modified) with permission form ref. 43 Copyright 2019 Royal Society of Chemistry. (b) Size control of MOFs – as an illustrative example, we present the downsizing of PCN-224 from a millimeter scale to the nanoscale which was achieved by diluting the system. This approach preserves the phase purity of the system, as it does not change the stoichiometry, while facilitating the creation of more MOF monomers – resulting in smaller particle sizes. Adapted (modified) with permission from ref. 47 Copyright 2016 American Chemical Society. (c) Size control of COFs: manipulating the concentration of a modulator facilitated size-control of a spherical COF. Adapted with permission from ref. 48 Copyright 2019 American Chemical Society.

monocarboxylic acids, are used to control crystal size and morphology. However, other functionalities, such as triethylamine, have also been employed in carboxylate-based MOFs.^{50,53}

(ii) pH mediated: higher pH levels shift linker deprotonation equilibria forward, speeding up nucleation rates and influencing morphology.^{49,50} For example, Wang *et al.* reported a change in the morphology of HKUST-1 particles influenced by pH adjustments.⁵⁴ By varying the amounts of sodium formate and triethylamine, they were able to change the pH from 2.23 to 3.88. This shift in pH altered the shape of the synthesized particles (Fig. 5b), transforming them from octahedrons to oval particles.⁵⁴

(iii) Surfactant: amphiphile surfactants adsorb onto crystal facets, altering morphology and controlling particle size (Fig. 5c). The degree of surfactant adsorption depends on its affinity for a specific crystal facet, which is determined by the

This article is licensed under a Creative Commons Attribution 3.0 Unported Licence.

Open Access Article. Published on 08 April 2025. Downloaded on 7/14/2025 10:04:51 AM.





Fig. 5 Shape-/morphology-controlled synthesis of reticular materials. (a) In a coordination modulation approach, the modulator and the linker compete for metal-ions. A strong control over these competitive interactions facilitates morphology control. Reprinted with permission from ref. 50 Copyright 2023 Elsevier. (b) Higher pH levels tend to shift the linker deprotonation equilibria toward the forward direction, accelerating nucleation rates and consequently affecting morphology. This approach is demonstrated here, for HKUST-1. Reprinted with permission from ref. 50 Copyright 2023 Elsevier. Data originally reported in ref. 54. (c) Surfactants exhibit a preference for attaching to specific crystal facets due to favorable interaction energies. Consequently, certain facets experience faster growth rates compared to others, primarily because these molecules either facilitate or hinder the attachment of reactant species to these facets. Reprinted with permission from ref. 55 Copyright 2023 Springer Nature. (d) Morphology control of COFs achieved by adjusting the solvent and inhibitor ratio: H₂O/PEG-400; (i) 0.5 mL/0.5 mL and (ii) 0.5 mL/1 mL. Reprinted with permission from ref. 56 Copyright 2022 Royal Society of Chemistry.

interaction energy. This selective adsorption can accelerate the growth rate of one facet while decelerating another, affecting how easily reactants can attach to the surface. As a result, surfactant addition influences the shape and size distribution of the final product. Additionally, amphiphiles can aggregate in water and certain solvents to form a soft matrix for nanostructures.⁵⁰ As an example, Lyu *et al.* used cetyltrimethylammonium bromide (CTAB) to produce truncated rhombic dodecahedra (TRD) and rhombic dodecahedra (RD) ZIF-8 particles.⁵⁵

(iv) Solvent-mediated: by adjusting solvent properties, crystal growth and shape can be controlled, affecting nucleation rates, precursor solubility, and crystal formation kinetics.⁵⁷ For example, for TAPB-DMTP COF,⁵⁶ varying PEG-400 to water ratios impacts the COF's morphology and crystallinity (Fig. 5d). At a ratio of $H_2O/PEG-400 = 0.5 \text{ mL}/2.5 \text{ mL}$, a strong diffraction peak at 2.79° disappeared, indicating disordered imine-linked condensates instead of well-defined COFs. With increasing PEG-400 content, SEM images revealed a shift from solid nanofibrous structures to chrysanthemum-like morphologies, highlighting the importance of solvent composition in tailoring COF properties for diverse applications without high temperatures or harmful solvents.⁵⁶

In their recent review, Zaleska-Medynska *et al.*⁵⁰ cover the strategies and mechanisms for controlling the morphology of

synthesized MOF particles. This review covers morphology control of porous materials in a broader context. However, it is worth noting that these mechanisms can still present challenges when aiming to achieve low-dimensional structures such as fibers or nanosheets.⁵⁸ Similar strategies have also been used for COFs. Furthermore, porous materials themselves can serve as highly effective templates for generating porous carbon materials with desired morphologies.⁵⁹ This approach facilitates the creation of diverse structures like carbon nanorods and graphene nanoribbons.

2.3. Templated synthesis

Controlling the geometric shape of reticular materials by regulating the bonding behaviour of building blocks is still difficult. A key approach in the synthesis of reticular materials involves the use of templated methods, which are typically categorized into two primary strategies: external templating and internal templating:⁶⁰

(i) External templating – more relevant to the focus of this review – uses templates to control the morphology and structure of reticular materials.⁶⁰ The premise of template-assisted synthesis is a relatively straightforward three-step process: the preparation of the template, the synthesis of the desired material on the template, and the subsequent removal of the template (Fig. 6a). Depending on the choice of template, these strategies can be further categorized as hard templating or soft templating – as discussed below.⁶¹

(ii) Internal templating, or using reticular materials as templates themselves, involves leveraging reticular materials to create other materials while maintaining their original morphology. This strategy allows for precise control over the size, composition, and structure of the derived materials.⁶⁰ For example, ZnO@ZIF-8 nanowires were synthesized using ZIF-8 as a template for growing ZnO, demonstrating the utility of MOFs in fabricating complex nanostructures.⁶⁵ However, since internal templating focuses more on structuring other functional materials rather than the synthesis of reticular materials themselves, it falls outside the scope of this review. Readers seeking more information on both external and internal templating methods are encouraged to explore additional resources.⁶⁰

In hard templating, the templates employed exhibit relative rigidity, essentially acting as structural scaffolds to facilitate the growth of nanostructures with morphologies that complement the template's configuration.⁶⁶ In the case of MOFs, two distinct avenues for harnessing templating strategies have been used. The first involves the employment of templates to guide the synthesis of MOFs with tailored morphologies,^{63,67} while the second entails the utilization of MOFs themselves as templates for the fabrication of complex nanoarchitectures.⁶⁸ In a recent review addressing hard templating strategies for MOFs,⁶⁹ Luque *et al.* categorize strategies into three distinct groups: (i) sacrificial, wherein templates are dissolved or removed after MOF synthesis; (ii) semi-sacrificial, characterized by templates that are neither entirely eliminated nor fully retained in the final MOF structure; and (iii) non-sacrificial,

where templates persist after synthesis, giving rise to template-MOF composite materials. Within sacrificial templating strategies, hard templates, such as alumina, silica, and polystyrene spheres, offer a more stable approach.^{70,71} These rigid scaffolds guide the growth of reticular materials into specific shapes.⁵⁸ For instance, PS@ZIF-8 composites can be produced by embedding polystyrene spheres in the synthesis mixture and then removing them, resulting in hollow ZIF-8 structures.⁷² Also, silica templates offer a finer degree of control over shape, porosity, and surface area and can be conveniently functionalized to enable the synthesis of multifunctional MOFs.⁶⁹ SBA-15 mesoporous silica has been used as a template to grow MOF-5, illustrating how template structures can significantly influence MOF formation and properties.73 However, MOFs need to exhibit resistance to hydrofluoric acid (HF) when employing silica as a sacrificial template, as HF is the sole viable etching agent for silica. For two-dimensionally grown MOFs, layered double hydroxides (LDHs), composed of positively charged layers interspersed with anionic regions between them, prove effective as sacrificial templates, as illustrated in the case of bimetallic Zn-Co ZIFs (Fig. 6b).⁶⁹ Following this strategy of taking advantage of the different chemical stabilities of MOF precursors, metal-organic assemblies (MOAs) have been used as a template for the synthesis of a hierarchical-pore MOF (H-MOF) (Fig. 6a).⁶² An example of semi-sacrificial templates is illustrated by metal oxides. In the presence of appropriate organic linkers, surface metal oxide entities transform, leading to the formation of corresponding MOFs.⁷⁴ Through the manipulation of linker quantities and reaction durations, the extent of this conversion can be regulated, ultimately yielding metal oxide-MOF composite materials. When a metal oxide composite is used as a template, the partial degradation of the metal oxide fraction of the template yields yolk-shell structures like Pd@ZIF-8.75 The most extensive category of templates is nonsacrificial. These templates encompass a wide array of materials, including polymers, MOF-based structures, silica templates (spherical and mesoporous), noble metals, metal oxides, CNTs, LDHs, and zeolites.

For hard templates, material-template interactions must be tailored. In this context, various strategies have been devised, including surface interactions with the template, the electrostatic assembly of negatively charged shell materials onto positively charged cores,76 and techniques such as chemical vapour deposition (CVD)⁷⁷ and atomic layer deposition (ALD).⁷⁸ The main limitation of hard templating is the availability of templates with suitable dimensions, shapes, and surface characteristics. Furthermore, the high associated costs often render such techniques impractical for large-scale applications. In contrast, soft templating strategies hold promise for creating hierarchically porous MOFs. However, the synthesis conditions for MOFs may not facilitate the self-assembly of soft templates. Additionally, achieving mesopore formation post-template removal poses challenges, mainly due to phase separation. Phase separation occurs when components in a mixture separate into distinct phases, potentially disrupting the uniform pore distribution in the final MOF structure.⁶⁷ To mitigate



Fig. 6 Templated synthesis of reticular materials. (a) H-MOF formed through *in situ* Metal Organic Assembly mediated templated synthesis. Adapted under CC 4.0 license permission from ref. 62. (b) A schematic illustration of two-dimensionally grown bimetallic ZIFs on a sacrificial LDH template. Reprinted with permission from ref. 63. Copyright 2018 Wiley. (c) A schematic illustration of a microemulsion-guided assembly strategy for the synthesis of a cerium-based hierarchically macro-microporous MOF. (i) P123 and F127 (triblock copolymers) act as co-stabilizers and form a columnar microemulsion with toluene as the oil phase. The assembly occurs (ii) spherical stacking and (iii)–(v) epitaxial growth. Reprinted with permission from ref. 64 Copyright 2022 American Chemical Society. (d) TEM image of a UiO-66-based MOF prepared from another MOF acting as a precursor template. Adapted under CC 4.0 license permission from ref. 62.

these challenges, certain strategies have been developed, such as the introduction of Hofmeister ions.^{67,79} This approach, termed salting-in ion-mediated self-assembly (SIMS), promotes the self-assembly of MOFs under relatively milder conditions.⁷⁹ Typically, this includes ionic surfactants, nonionic copolymers, and emulsions (Fig. 6c).^{64,80,81} Soft templates, such as CTAB and triblock copolymers, are employed to create hierarchical porous MOFs by guiding the structure during synthesis. For example, CTAB has been used effectively to synthesize mesoporous MOFs with hierarchical porosity. However, the stability of small molecular micelles can be a limitation of this approach. In the case of using MOFs as templates for the synthesis of secondary MOFs, the approach leads to the generation of hierarchically porous structures with mesoporous sizes controllable by regulating template quantity (Fig. 6d).⁵⁸ Conversely, template MOFs may serve as seeds for the epitaxial growth of secondary MOFs with new linkers, leading to core–shell MOF composites.^{58,82} Again, these principles of

templating have been further translated to the structuring of other classes of porous materials like COFs⁸³⁻⁸⁶ and POCs.⁸⁷

2.4. Self-assembled superstructures

Porous reticular materials have been assembled as 0, 1, 2 and 3D-ordered superstructures. 0D superstructures typically form in tightly confined nano- or micro-spaces, often through methods such as solvent evaporation or spray drying.⁸⁸ For example, Vogel et al. used solvent evaporation from the surface of emulsion droplets to achieve diverse morphologies of ZIF-8 (such as cubic, rhombic dodecahedral, and truncated rhombic dodecahedral) and octahedral UiO-66 as self-assembled superstructures (Fig. 7a).⁸⁹ Moving to 1D superstructures, induction of 1D self-assembly of particles often necessitates the use of driving forces such as electric and magnetic fields. Granick et al. capitalized on the surface polarization of the electrostatic double layer of ZIF-8 particles, dispersing them in ethylene glycol and applying an AC electric field to generate 1D chains.⁹⁰ This assembly arose from particle-to-particle contact driven by dipole-dipole interactions (Fig. 7b). Formation of 2D superstructures has been done through the gradual evaporation of colloidal solutions of MOF particles (Fig. 7c),⁹¹ the spreading of colloidal solutions across air-liquid interfaces with subsequent interface modification to drive assembly,⁹² and the surface modification of MOFs using oligonucleotides, capitalizing on associated hydrogen bonding.⁸⁸ In turn, the creation of 3D superstructures has been achieved through sedimentation methods,⁹² solvent evaporation⁹³ and the use of specifically tailored surfactants (Fig. 7d).94 Moving to COFs, relatively few comprehensive studies have focused on mechanisms governing self-assembly. An early investigation in this context was from Banerjee et al.,⁸⁴ who studied the self-assembly mechanisms of hollow-sphere COFs. Their approach involved isolating samples at fixed intervals during the initial reaction and subjecting them to detailed analyses using scanning electron microscopy (SEM), transmission electron microscopy (TEM) and atomic force microscopy (AFM). At 12 h, they found rod-like morphologies, with crystallites forming through π - π stacking of COF layers. As the reaction progressed to 24 h, these crystallites began to aggregate, forming a spherical morphology with hollow cavities forming inside after 36 h of reaction time. At 48 and 72 h, the surfaces of these structures smoothened out due to the fusion of crystallites. Similar in-depth analyses have been done to study the self-assembly mechanisms governing microtubular COFs.95

3. The macroscopic structuring landscape

In the last ten years, there has been a big push in research to figure out and use different ways to shape and structure porous materials such as MOFs.⁹⁶ Here, we discuss the various shapes and structures one can use to get the best performance in industrial settings. The central idea is to keep the material's core properties intact from the tiny, microscopic level all the



Fig. 7 The self-assembly of superstructures. (a) Solvent evaporation from the surface of emulsion droplets results in the formation of a rhombic dodecahedral 0D ZIF-8 superstructure. (i) SEM image of monodisperse particles, (ii) SEM image of assembled superstructure. Reprinted with permission from ref. 89 Copyright 2022 Wiley. (b) Surface polarization of the electrostatic double layer leads to the assembly of 1D chains of ZIF-8 crystals. (i) A schematic illustration of the process, (ii) confocal cross-sections perpendicular to the applied field. Reprinted with permission from ref. 90 Copyright 2013 American Chemical Society. (c) Formation of 2D superstructures achieved through the gradual evaporation of MOF particle colloidal solutions. SEM image of the superstructure (scale = 1 μ m). Reprinted with permission from ref. 91 Copyright 2012 Wiley. (d) Surfactant-mediated orientation results in 3D MOF superstructures. (i) and (ii) SEM images depict the packing of MOF crystals in the superstructure (scale = 2 μ m, 1 μ m respectively). (iii) Schematic illustration of packing. (iv) SEM image of the crystal. Reprinted with permission from ref. 94 Copyright 2019 Wiley.

way up to the larger, macroscopic scale to ensure it works just as well in real-world applications. With a whole array of macroscopic forms to consider - such as granules, pellets, self-assembled monoliths, thin films, gels, foams, and glasses - picking the right shape gets more complex. The choice of shape is largely dictated by the application requirements. For instance, in gas storage applications, given fast kinetics, the most important metric is - arguably - the volumetric capacity of the adsorbent. Here, powdery materials with low bulk density will perform poorly, whereas dense monoliths will be the go-to choice. Moreover, due to the nature of the application, the adsorbent is subject to a high degree of mechanical stress during handling and operation, demanding high mechanical strength. In such cases, monoliths fabricated through a sol-gel route are promising candidates since they provide the native porosity of the porous structure but facilitate excellent mechanical properties and particle densities close to single-crystal ones.18 In molecular sieving-based applications - such as gas separations and water decontamination⁹⁷ – pore sizes, chemical stability, tensile strength and flexibility are of primary importance wherein membranes can be ideal candidates33,98 - although high density is still of major importance.

Broadly speaking, the industrial applications covered here are directly linked to the adsorption properties of the porous materials, and thus their performance will be dictated by properties such as adsorption uptake, selectivity, adsorption kinetics, heat of adsorption, heat conductivity, and stability with cost being another important question. Importantly, the choice of shaping process has an impact on these properties. For example, the use of binders might partially block the porosity and reduce the adsorption uptake,⁹⁹ while the application of mechanical pressure may partially collapse the porosity. Adsorption kinetics are particularly relevant and often ignored since, again, the shape factor is critical. Here, techniques leading to thicker-shaped materials might result in longer diffusion pathways, which would reduce the efficiency of the material's performance. At the same time, a lack of densification might reduce the density of the final material and, hence, the volumetric adsorption capacity as well as the heat conductivity, key for heat dissipation during the exothermic adsorption process.¹⁰⁰ Looking back at binders, a poor selection might result in the creation of microstructures with reduced thermal conductivity compared to the original bulk material.

Stability is another crucial consideration – which has been the focus of several excellent reviews^{33,101} – and thus is discussed in the present context very briefly. Here, stability takes different forms – mainly thermal, chemical, and mechanical. The ability to withstand relatively high temperatures without undergoing significant structural or property changes is crucial for the success and reliability of shaped materials.^{102,103} Chemical stability is the ability of shaped materials to remain chemically stable in the presence of reactive gases, corrosive substances, acidic or basic environments.^{102,104} Mechanical stability ensures that the shaped materials can withstand mechanical stresses due to *e.g.*, moves, vibrations, and the weight of a packed bed and maintain their structural integrity during handling, installation, and use, ultimately enhancing their reliability and longevity in real-world applications.

Although the underlying principles governing reticular material stability are universal, application-specific considerations need to be investigated. We present here a discussion on various shapes including granules, pellets, monoliths, sol-gel monoliths, foams and gels, thin films, and glasses (Fig. 8). In turn, Section 4 will detail the structuring techniques utilized for their formation, providing case studies that illustrate the implementation of structuring techniques to produce reticular materials.

3.1. Granules

Granular materials (Fig. 8a) are generic shapes used in industry as they are relatively easy to translate to large scales at relatively lower costs.¹¹² These solid bodies come in various sizes and associated particle size distributions,¹¹³ Their flexibility facilitates the customization of these structures to meet the requirements of different processes and industries. In particular, granules are used in packed beds and fluidized bed reactors, where the particle size distribution enhances heat and mass transfer.¹¹⁴ At the same time, the particle size and packing density of the granules impact their flowability,¹¹⁵ i.e. the ability of the powder or granules to fall or flow over the influence of their weight. In general, high flowability is desired because it eases the handling and processing of the materials, something that is achieved using large particle sizes. On the other hand, large particle sizes reduce the adsorption kinetics and, therefore, the optimal granule size depends on the final engineering system. Typically, these structures must withstand immense pressures and associated crushing forces throughout their application and, therefore, their mechanical properties are critical. Crucially, granules can retain the porosity of the material obtained at the microscopic scale, with acceptable levels of drops in the porosity upon structuring.

MOFs shaped as granular composites have demonstrated good capabilities in terms of adsorption performance. For example, granular composites of MIL-100(Fe) loaded onto an alumina support demonstrated enhanced performance for tetracycline hydrochloride adsorption when compared to bare MOF powder.¹¹⁶ The composite exhibited an adsorption efficiency of 95% within 60 minutes, even with low MOF loading (approximately 3 wt%), compared to both activated Al₂O₃ alone (51%) and MIL-100(Fe) powder alone (72%). Additionally, the results revealed a broad pH applicability range (pH 4 to 10) for the composites, with minimal influence from most inorganic ions in solution, apart from fulvic acid and carbonate. Moreover, the composites displayed excellent operability, recyclability, and regenerability through photolysis. The superior adsorptive performance of the granular MOF composites was attributed to synergistic interactions between the MOF layer and activated Al₂O₃ support. The successful loading of MIL-68(Al) onto activated Al₂O₃ further confirmed the efficacy of the synthesis route.116



Fig. 8 Different macroscopic forms of reticular materials: (a) granules: MIL-88B(Fe) granules, reprinted with permission from ref. 105 Copyright 2022 Elsevier. (b) Pellets: HKUST-1 pellets, reprinted with permission from ref. 106 Copyright 2020 MDPI. (c) Sol-gel monoliths: UiO-66 monoliths formed using a sol-gel route, reprinted with permission from ref. 107 Copyright 2019 Springer Nature. (d) Foams and gels: MOP/COF composite aerogel, reprinted with permission from ref. 108 Copyright 2024 Springer Nature. (e) Thin-films: wrinkled MOF thin-film with Turing patterns on a porous alumina support, reprinted with permission from ref. 109 Copyright 2024 American Association for the Advancement of Science. (f) Monoliths: MIL-101(Cr) monolith structured using an extrusion-based method, reprinted with permission from ref. 110 Copyright 2024 Springer Nature. (g) Glasses: schematic visualisation of a ZIF-8 glass, reprinted with permission from ref. 111 Copyright 2024 Springer Nature.

3.2. Pellets

Pellets are small, typically cylindrical, or spherical solid particles (Fig. 8b).¹⁰⁶ Compared to granules, pellets have narrower particle size distributions, achieved through more precise manufacturing techniques, typically extrusion and spheronization. Sometimes, MOF powders can be tableted, providing morphologies similar to pellets.¹¹⁷ Uniformity in size and shape is particularly important in packed bed configurations, as it ensures consistent flow patterns and minimizes pressure drop variations.¹¹⁸ More than granules, pellets are engineered for high mechanical strength and crush resistance, essential for withstanding the rigors of handling, transport, and use in industrial settings. Pellets can endure high mechanical pressures and stresses, ensuring they maintain their structural integrity throughout the duration of a process.¹¹⁹ For example, pellets of HKUST-1 have been prepared using extrusion - an industrially scalable process - where water is added to the MOF powder to form a paste, which is then heated in a syringe (Fig. 8b).¹⁰⁶ Using sucrose as a binder, a Zr-based MOF powder has been shaped into spherical pellets with diameters ranging 0.5 to 15 mm in a kilogram-scale batch.¹²⁰ These Zr-MOF pellets showed no mechanical degradation after 70 consecutive drops from a 0.5-meter height, while approximately 5% of the pellets fractured during a 60-minute tumbling test at 25 rpm. In this example, however, it should be noted that there was a nearly 50% reduction in surface area, resulting in just 60% of the original hydrogen storage capacity of the powdered MOF (*i.e.* 1.54 wt% measured at 77 K and 1 bar). The volumetric capacities of the pellets cannot be compared as there is no density reported for the synthesized and shaped materials.¹²⁰

3.3. Monoliths

The IUPAC defines a monolith as a shaped, fabricated body with a homogenous microstructure that does not exhibit any structural components that can be distinguished by optical microscopy.¹²¹ Going by this generic definition, many shapes being discussed in this review can be considered monoliths. In industry, however, monoliths are seen as structured beds featuring large channels – straight or a labyrinthine network (Fig. 8f). The channels' size and connectivity are essential as they promote fluid and reactant flow and ensure efficient mass transport while minimizing flow resistance and pressure drop. Monoliths provide high pore volume and external surface area, essential to promote active sites for adsorption and catalysis.¹²² Typical examples include honeycomb monoliths used in catalytic converters. Here, channel connectivity enables reactant access to catalytic sites, influencing reaction rates.¹²³ Monoliths enable fast kinetics and are engineered for structural stability and mechanical strength - characteristics vital for long-term use in the demanding environments where they are used. They can withstand variations in pressure, flow rates, and temperature, making them suitable for continuous processes in industries such as petrochemicals and environmental remediation.¹²⁴ Importantly, a monolithic, structured bed can be coated by an active phase for adsorption or catalytic activity. As an example, Darunte et al. demonstrated that honeycomb monoliths supported with mmen-Mg₂(dobpdc) exhibited a CO₂ uptake of 2.37 mmol g⁻¹ and excellent cyclic adsorption/ desorption performance.¹²⁵ The monoliths had MgO and Mg₂(dobpdc) loadings of 8-10% and 14-18%, respectively. Despite being tested on 100 cells per square inch (cpsi) monoliths, it is anticipated that utilizing dimensional differences could lead to 2-3 times higher weight loading on 400 or 600 cpsi monoliths. Further optimization could result in different MgO loadings on the monolith, affecting the transformation rate from MgO to Mg₂(dobpdc). Nitrogen isotherms showed significant microporosity after growth, with N2 physisorption uptake reaching 58.5% of Mg₂(dobpdc) in powder form at $P/P_0 = 0.6$. CO₂ adsorption capacity was 2.37 mmol g^{-1} at 10% \mbox{CO}_2 concentration in helium and 2.88 mmol g^{-1} using pure CO₂, with film-form mmen-Mg₂(dobpdc) achieving approximately 70% of powder-form uptake at 10% CO₂ concentration.¹²⁵

Along similar lines, Lawson et al.¹²⁶ investigated the immobilization of MOF-74(Ni) and UTSA-16(Co) on commercial cordierite monoliths (600 cpsi) for CO₂ capture. MOF-74(Ni)and UTSA-16(Co)-cordierite monoliths with loadings as high as 52% and 55%, respectively, were prepared using layer-bylayer + secondary growth and in situ dip coating techniques. Both methods produced uniform MOF layers on the cordierite surface. The layer-by-layer plus secondary growth method showed promise for MOF-74(Ni) growth, while in situ dip coating yielded thick UTSA-16(Co) layers. They further improved this process by incorporating the MOFs on a polyamide-imide Torlon monolith using the existing carbon hollow fiber surface. Prior to MOF growth, the carbon hollow fibers were functionalized with hydroxyl groups to improve their integration. They then used dip-coating and layer-bylayer techniques to grow MOFs. The composites exhibited loadings of 38 wt% of MOF and film thicknesses ranging 10 to 15 mm and surface areas of 266 and 211 m² g^{-1} for MOF-74/ carbon and UTSA-16/carbon composites, respectively, along with pore volumes of 0.28 and 0.20 $\text{cm}^3 \text{ g}^{-1}$. As proof of concept, they reached CO2 adsorption capacities of 1.2 and 2.0 mmol g^{-1} for MOF-74 and UTSA-16 composites, respectively, at room temperature and 1 bar. Looking ahead, alternative methods may be explored to refine the coating procedure and thereby, improve film growth on carbon hollow fibers, with the goal of mitigating the reduction in BET area associated with strong acids.126

3.4. Sol-gel monoliths

In addition to the conventional definition, the term 'monolith' is used many times to describe a single, continuous structure without joints or seams, often characterized by a uniform material composition throughout (Fig. 8c). This definition emphasizes the monolithic nature as a single piece. The term 'monolith' describes, therefore, single bodies of shaped materials formed during synthesis, which can lead to either densified materials, where the adsorbent's volumetric capacity is crucial¹²⁷ but also ultralight ones.^{128,129} These shaped materials often resemble foams, aerogels and xerogels described in Section 3.5, and are, frequently, referred to by these names in the literature. Arguably, foams and gels can be monoliths but only receive this name when they are structured as a single body rather than a powder.⁹⁹ Monolithic adsorbents are, for example, used in chromatography and benefit from their large surface area and macro-/mesoporous network, enabling efficient separation of compounds in complex mixtures.¹³⁰ Over the past few years, various analogues of sol-gel monolithic MOFs and COFs have been synthesized and reported in the literature¹²⁷ (Fig. 9), beginning with the ZIF-8 monolithic synthesis reported by Tian et al. in 2015.131

Sol-gel materials have been studied for a range of energyrelated applications, particularly in gas storage, where they have displayed record performances due to their high density and volumetric adsorption capacities.^{16,18} A notable example of such monoliths in gas storage applications is the use of HKUST-1 for H₂ storage.¹⁸ The relatively high density of monoliths enhances their volumetric adsorption properties, rendering them advantageous for diverse uses. Unlike conventional structures, sol-gel monoliths do not require binders in their formation, resulting in a more uniform and pure material composition. An additional advantage lies in the precise control over their micro/mesoporosity achievable by manipulating synthesis parameters.¹⁰⁷ This level of control facilitates the creation of tailored pore structures, allowing for the fine adjustment of properties to suit specific application needs.

3.5. Foams and gels

Foams consist of empty macropores encased within a denser matrix or skeleton. In contrast, a gel is a non-fluid, colloidal network saturated with a liquid. When the space within a gel is evacuated and replaced by a gas, preventing minimal structural change, it transforms into an aerogel (Fig. 8d). Conversely, reducing this space results in the formation of a xerogel.¹³² Based on their composition, cell morphology, and physical properties, polymer foams can be categorised into two types: rigid or flexible. Depending on the size of the foam cells, they can be further classified as macrocellular (larger than 100 µm), microcellular (ranging from 1 to 100 µm), ultra-microcellular (0.1 to 1 µm), and nanocellular (0.1 to 100 nm). Furthermore, polymer foams can be classified as either open-cell or closedcell. Closed-cell foams are characterised by isolated voids and cavities surrounded by cell walls. In contrast, open-cell foams possess broken cell walls and primarily consist of ribs and



Fig. 9 Chronological overview of sol-gel monoliths, accompanied by optical images, illustrating the self-shaping evolution of key monolithic MOFs and COFs. Reproduced with permission from ref. 127 Copyright 2023 Wiley.

struts.¹³³ The defining feature of both foams and gels is their high porosity, marked by a substantial volume of voids or pores.¹³⁴ For instance, Chen et al. demonstrated a continuous phase transformation strategy to shape MOFs from processable fluids (Fig. 10).¹³⁵ They presented a cup-shaped Cu-MOF composite based on HKUST-1 and a hierarchically porous MOF foam with high catalytic efficiency in C-H oxidation, achieving 76% conversion and 93% selectivity for the composite, and 92% conversion and 97% selectivity for the foam, along with improved recycling and kinetics. This procedure was extended to ZIF-8, Mg-MOF-74, Zn-MOF-74, UiO-66, and NH₂-UiO-66. These low-density MOF-based foams ($<0.1 \text{ g cm}^{-3}$) with high MOF loadings (up to 80 wt%) featured hierarchically porous structures and uniformly distributed, fully accessible MOFs. These foams exhibited a low energy penalty (pressure drop <20 Pa at 500 mL min⁻¹) and, as a result, can be employed as membrane reactors.

3.6. Thin films

A thin film is a layer that extends in two dimensions but the thickness ranges from several nanometers to a few micrometers and are designed for use as coatings on a substrate (Fig. 8e). Their thickness is controlled to meet specific needs; precision in this context is vital, as slight variations in thickness can have an important influence on their performance.¹³⁶ Moreover, thin films must maintain uniformity across the entire surface to ensure consistent functionality. Additionally, the surface of thin films can be finely tuned to achieve controlled roughness, which impacts properties such as adhesion, friction, and wettability, thereby facilitating tailored interactions with other materials.¹³⁶ For instance, Luo *et al.* introduced an innovative

synthesis method to produce wrinkled thin films of HKUST-1 with Turing patterns (Fig. 8e), successfully balancing high MOF loading with mechanical flexibility.¹⁰⁹ By employing a confined interfacial reaction between a zinc oxide thin film and a polymer topcoat, they achieved wrinkle configurations that enhance strain tolerance up to 53.2% and significantly increase the surface area, reaching a BET area of 1473 $m^2 g^{-1}$. The films displayed a notable MOF loading of 96.3 wt% and formed 13 distinct Turing patterns through precise adjustments of reagent concentrations and polymer thickness. In practical applications, these films demonstrated impressive hydrogen permeance of up to 6.82×10^4 gas permeation unit (GPU) and H₂/CO₂ selectivity of 15.3. When stacked, the films achieved even higher selectivity (41.2) and permeance (8.46 \times 10³ GPU). Furthermore, the films maintained their structural integrity during transfer onto various substrates and exhibited a high elongation at break of 41.6%, highlighting their potential for use in flexible electronics and efficient gas separation membranes.109

Surface roughness and morphology can also impact sensing and biosensing performance. For example, Chen *et al.* fabricated on-chip electrochemical micro-biosensors using an electrically conductive Cu-benzenehexathiol (Cu-BHT) film (Fig. 11a), with a flat upper surface (Fig. 11b) and an undulating bottom surface (Fig. 11c).¹³⁷ The bottom surface, with dense crystal defects (ts-Cu) acting as nanozymes, showed higher H_2O_2 sensing performance than the smoother upper surface. Crystal defects can enhance the electrocatalytic interfaces, typically buried between the solid support and liquid electrolyte in conventional sensing methods. Thin films can be tailored based on the material used and so can be engineered



Fig. 10 Shaping different MOF structures into foams. Foams are materials containing gaseous voids surrounded by a denser matrix. Some gel stages can also be technically classified as foams. The defining feature of such structures is high porosity, characterised by a significant volume of pores and voids. To illustrate the concept of these structured materials, we present (i) and (ii) the precursors, (iii) crystal structures, (iv) optical images, and (v) and (vi) SEM images with a scale of 100 and 1 μ m, respectively, for (a) HKUST-1, (b) ZIF-8, (c) Mg-MOF-74, (d) Zn-MOF-74, (e) UiO-66, and (f) NH₂-UiO-66. Adapted with permission from ref. 135 Copyright 2016 American Chemical Society.

with precision to achieve specific pore size distributions, typically within the range of micropores.¹³⁸ This is of course relevant for applications such as filtration or controlled diffusion, where selective separation and diffusion is a primary requirement.^{139–141}

3.7. Glasses

While the discussion thus far has predominantly centred on solid, crystalline reticular materials, recent explorations into the order–disorder dynamics and solid–liquid–glass transitions within certain MOFs have shown the possibility of creating glassy, non-crystalline materials (Fig. 8g). Here, it is important to draw a distinction between amorphous MOFs and MOF glasses. Ma and Horike define a 'glass' as an 'undercooled frozen-in liquid'.¹⁴² For an amorphous MOF to be classified as a 'glass' according to this definition, there needs to be an absence of long-range atomic ordering and a second-order

phase transition to a soft liquid-like state at the glasstransition temperature (T_{α}) . For more theoretical insights, we refer readers to focused reviews dedicated to MOF glasses.^{142,143} In principle and given that a material does not lose its intrinsic properties such as porosity, melting a material is potentially the most effective way of implementing shaping and densification. However, and despite their recent interest, the transition from crystalline to glassy phases often results in the loss of the innate porous structure of coordination polymers,³⁴ limiting its impact in conventional applications such as gas storage, separation, and catalysis. Consequently, glassy materials may face challenges when competing with porous counterparts due to the lack or reduced porosity.¹⁴⁴ On the other hand, glassy MOFs - like glassy carbons before - lack grain boundaries, enhancing their mechanical stability and, for conductive MOFs, their conductivity, while their flexibility enables easier manipulation. This makes



Fig. 11 (a) A schematic illustration of the gas–liquid interfacial reaction method utilised to produce Cu-BHT films. (b) A schematic of the structure of the prepared Cu-BHT film. (c) Optical microscopy images of the prepared film. (d) A schematic illustration of the transfer method for obtaining the flat upper-side surface. (e) AFM images of the flat upper-side surface. (f) A schematic illustration of the flat upper-side surface of the Cu-BHT film. (g) A schematic illustration of the transfer method for obtaining the bottom-side surface. (h) AFM images of the bottom-side surface. (i) A schematic illustration of the bottom-side surface. (i) A schematic illustration of the bottom-side surface. (ii) A schematic illustration of the bottom-side surface. (iii) A schematic illustration of the bottom-side surface. (iii) A schematic illustration of the bottom-side surface. (ii) A schematic illustration of the bottom-side surface. (iii) A schematic illustration of the bottom-side surface. (iiiii) A schematic illustration of the bottom-

them good candidates for various applications, particularly as electrodes,¹⁴⁵ electrolytes,¹⁴⁶ or membranes.¹⁴⁷

4. Structuring of reticular materials at the macroscopic scale

While there have been several developments with regards to 'novel' techniques for achieving the shapes introduced so far, there are complexities that need to be considered for industrial translation. These considerations are many times related to aspects of process engineering more than to the chemistry of the materials. Important considerations include production throughput – *i.e.*, the amount of product that is produced in terms of the space-time yield,⁹⁶ 'quality' of the final product, and cost of the overall process – *i.e.*, cost of reagents, human power, utility, and transportation, among others. In most applications, it is necessary to reach a compromise involving these aspects. For instance, in applications where products of

high-quality may be required (e.g., sensing), the production throughput may be low, and the costs may be high. Conversely, applications where high throughput and low cost is desired may suffer from lower product quality. In this section we offer insights into the techniques available for achieving the diverse landscape of shapes discussed in Section 3. We classify these techniques into two major categories as in situ shaping and post-synthetic shaping. in situ shaping techniques involve the synthesis and structuring of materials simultaneously in a single step, offering precise control over the final shape and structure. Examples include sol-gel synthesis, dip-coating, deposition techniques, and spray drying. Conversely, postsynthetic shaping techniques involve shaping of the already synthesized materials, mostly in the powder form. This category encompasses methods like pelletization, granulation, extrusion, spheronization, 3D printing, phase inversion, hydrogelation, and glass formation, allowing greater flexibility in shaping materials after synthesis, thereby enabling the production of complex geometries and tailored structures.

4.1. In situ shaping

4.1.1. Sol-gel monolith synthesis. From a mechanistic standpoint, the formation and packing of the primary particles of a sol-gel during the synthesis procedure governs the macroporous structure and macroscopic properties.¹²⁷ An example of this is shown in Fig. 9 and 12a, with cm-sized monoliths. These monoliths result from the very small NPs formed during the sol-gel process, contributing to the formation of a gel product (Fig. 12b and c). Selfassembly of primary particles in sol-gel monoliths occurs through a sequence of well-defined processes, including supersaturation, nucleation, growth, and Ostwald ripening (Fig. 12d).127,148 During the nucleation stage, the MOF seeds, or nuclei, act as substrates for subsequent crystal growth. During the growth stage, the interplay of external surface reactions and diffusion of monomers becomes important. During the subsequent Ostwald ripening stage, larger particles grow at the expense of smaller particles.⁴⁰ Finally, during the drying process of the sol-gel synthesis, the gradual solvent removal increases the likelihood of nanoparticle agglomeration

into bulk monolithic structures.⁹⁹ Overall, controlling particle size during sol–gel synthesis is crucial for the formation of monolithic structures, emphasizing the importance of employing appropriate techniques for size control during the sol–gel reaction to yield optimal structures.¹²⁷

Controlling the kinetics of material formation is a fundamental method for regulating particle growth. By shortening the reaction time, starting crystals have less time to grow, resulting in smaller primary particles. However, this can result in a lower yield and consequently, reduced crystallinity and surface area due to unreacted precursors.¹²⁷ Temperature is another crucial factor in crystallization kinetics; lowering the synthesis temperature slows down the reaction, resulting in smaller primary particles.^{151,152} However, altering temperature can have complex effects on the chemical reaction. Additionally, not all materials can be synthesized at lower temperatures, as some reactions require a minimum temperature to proceed. For instance, while ZIF-8¹³¹ and RT-COF-1¹⁵³ can be synthesized at room temperature, UiO-66 requires 100 °C.¹⁰⁷ Upon



Fig. 12 The self-assembly of superstructures. (a) The image showcases the solid and mechanically sturdy macrostructure of a Zr-based MOF, UiO-66- NH_2 , obtained through the supercritical CO_2 drying technique from the initial gel state. This macrostructure effectively retains the intricate features and inherent characteristics of the microstructure, as evidenced by the SEM micrographs presented in (b) and (c). Reprinted with permission from ref. 149 Copyright 2018 Royal Society of Chemistry. (d) A schematic representation of MOF NP nucleation and growth based on the LaMer model. The blue trace illustrates that the synthesis of uniform small MOF NPs involves a rapid formation of numerous, abundant nuclei. In contrast, the red trace shows that a limited number of nuclei and a slower growth rate lead to the formation of uniform large NPs. Reproduced with permission from ref. 150 Copyright 2018, John Wiley and Sons.

achieving a gel of primary particles, obtaining a monolith involves several steps. Firstly, centrifugation is used to separate the material from the reaction media. The speed at which centrifugation is performed plays a critical role in the formation of the monolith. Higher centrifugation speeds lead to stronger centrifugal forces acting on the particles, causing them to settle more rapidly. Consequently, faster centrifugation rates typically result in better compaction of primary particles, leading to denser monoliths. This is because the increased centrifugal force helps to pack the particles more closely together, reducing the void spaces between them. As a result, the final monolith exhibits enhanced structural integrity and mechanical strength. Conversely, lower centrifugation speeds may not provide sufficient force to compact the particles effectively, resulting in a less dense and weaker monolith.¹²⁷

Following centrifugation, the drying phase emerges as a critical step, demanding customized approaches tailored to each material type.127 Typically, slower drying processes are preferable for monolith formation over powder formation. Material-specific considerations come to the forefront during this stage; while for many MOFs may suffice with room temperature drying, rigid COFs often necessitate supercritical CO2 $(scCO_2)$ drying to avoid the meniscus and capillary forces in the gas-liquid interface when evaporating the solvent. For example, in the case of TPB-DMTP-COF pellets, higher acetonitrile fractions result in a sharp decline in the BET area due to pore disruption induced by capillary action. To address this issue, a sample processed in pure acetonitrile and activated in scCO₂ not only restored full porosity but also exhibited a BET area consistent with the desired trend. This offers crucial insights into the efficacy of scCO₂ treatment in mitigating pore damage and preserving porosity and surface area. Moreover, fine-tuning of the BET area was achieved by adjusting the rate of scCO₂ pressure release, further affirming the importance of optimal supercritical activation in maintaining material integrity.¹⁵⁴ Another instance of material-specific treatment is evident in Cu-centered MOF (monoHKUST-1) monolith formation, where the drying temperature is dictated by particle size. Larger particles necessitate lower drying temperatures to achieve a monolith, while smaller ones can withstand higher temperatures for monolith formation. For example, particles with a size of 51 nm can be dried at 40 °C to achieve a monolithic structure, whereas those with a size of 73 nm can only endure drying at 30 °C. Conversely, particles sized at 145 nm fail to yield a monolithic structure at any temperature, attributed to mechanical stress induced by solvent surface tension.¹⁸ A similar strategy allowed us to produce monoliths for Zr-based UiO-66 (monoUiO-66) MOF.¹⁰⁷

One of the key challenges in advancing adsorption technologies lies in the hesitation to report true volumetric adsorption capacities based on real – rather than single crystal – densities. This reluctance is influenced by several factors. Firstly, measuring density, with multiple definitions including skeleton, envelop and bulk, is inherently challenging, complicating the experimental process. Additionally, there is a common lack of awareness among researchers about the significance of volumetric data. However, understanding the importance of sol-gel processes in densifying MOFs and COFs sheds light on the significance of such measurements. By avoiding high mechanical pressures, these processes yield materials with exceptional volumetric adsorption performances. Indeed, while some MOFs may not be optimal with respect to mechanical stability, and cannot be shaped using mechanical pressure due to pore collapse, they can be effectively shaped and densified using a sol-gel method. For example, while HKUST-1 tablets prepared under 100 bar pressure shown a reduced BET area of 600 m² g⁻¹, robust HKUST-1 monoliths showed a gravimetric BET area of 1550 m² g⁻¹, much higher than via mechanical pressurization.¹⁵⁵ Not only the gravimetric BET area is affected but the volumetric shows the impact of densification, with BET areas of 925 m² cm⁻³ for the powder compared to 1651 m² cm⁻³ for the monolith.¹⁵⁶ In the case of ZIF-8, monoZIF-8 showed a gravimetric BET area comparable to its powder equivalent (ca. 1400 m² g⁻¹).¹³¹ However, considering the density of the monolith versus that of the powder, the volumetric BET area of the monolith reached 1660 m² cm⁻³ vs. 485 $m^2 cm^{-3}$ for the non-densified powder. This metric not only serves as a favorable indicator of dense monolithic material performance but also offers a more relevant measure of gas adsorption potential in real-world applications. While volumetric capacity might seem inconsequential in laboratory settings, it holds significant value in large-scale industrial applications and not only in automotive. Higher volumetric capacities facilitate a reduction in the footprint of adsorption systems, ultimately lowering costs. Thus, despite the challenges associated with measuring density, recognizing the practical implications highlights the importance of reporting real volumetric adsorption capacities.

The storage of natural gas, particularly through methane adsorption, has been a key focus in the development of highperformance porous materials. Among these, monoHKUST-1 and monoUiO-66 have demonstrated exceptional methane storage capacities, reaching 259 and 211 cm^3 (STP) cm^{-3} respectively, at 65 bar and 25 °C (Fig. 13a and b).18,107 Notably, monoHKUST-1 became the first porous material to meet the U.S. Department of Energy (DOE) target for methane storage, marking a significant milestone in this field. By comparison, conventional HKUST-1 powders, which had one of the highest methane storage capacities among pristine MOFs, exhibited a significantly lower total methane storage capacity, reaching only 185 cm³ (STP) cm⁻³ g⁻¹ L⁻¹ at 65 bar and 298 K (Fig. 13a). This value was achieved through pelletization at 27.6 MPa as a structuring technique.157 However, advancements in material engineering have since led to the development of dense sol-gel monoliths, which have surpassed these storage capacities. More recently, in hydrogen storage, monoHKUST-1 demonstrated an impressive capacity of 46.0 g L^{-1} at 100 bar and 77 K.¹⁵⁶ These results demonstrate the potential of monolithic porous materials in improving gas storage technologies and meeting energy targets.

In addition to densification, sol-gel, self-shaped monoliths typically display higher mechanical properties, surpassing the



Fig. 13 Monolithic MOFs exhibit volumetric BET areas that are three times higher compared to their powdered form. Additionally, they exhibit exceptional mechanical properties – surpassing the elastic modulus and hardness of single crystals. They also exhibit superior chemical stability. $_{mono}$ HKUST-1 displayed a remarkable methane volumetric adsorption of 259 cm³ (STP) cm⁻³ at 65 bar (a) the absolute volumetric methane adsorption isotherms at 298 K on monoliths and pellets of HKUST-1. $_{mono}$ UiO-66 achieved 296 cm³ cm⁻³ at 100 bar, a value comparable to that of $_{mono}$ HKUST-1 under high-pressure conditions. (b) Comparison of experimental isotherms for absolute volumetric CH₄ uptake at 298 K in $_{mono}$ UiO-66 and $_{mono}$ HKUST-1. Reproduced under the terms of the CC-BY 4.0 licence from ref. 127 The U.S. DOE volumetric CH₄ storage target of 263 cm³ (STP) cm⁻³ (65 bar) is indicated by the dashed red line.

elastic modulus - up to more than twice - and hardness - up to 30% - compared to single crystals.¹²⁷ Regarding the hydrochemical stability, monoZIF-8 retained its crystalline structure and monolithic morphology after being submerged in boiling water for 7 days.¹³¹ This additional stability is an opportunity to use sol-gel monoliths in more demanding conditions. For example, we have studied three monolithic Zr-MOFs for water adsorption: monoUiO-66, monoUiO-66-NH2, and monoZr-Fumarate.¹⁵⁸ Looking at the best overall performer, monoZr-Fumarate showed similar gravimetric BET area (854 $m^2 g^{-1}$) than the powder but far superior volumetric BET area, with 1063 and. 649 m^2 cm⁻³ for the monolith and powder, respectively. Notably, the three monolithic materials showcased superior volumetric water uptake performance at 25 °C and 90% RH compared to their powdered counterparts, with 0.30, 0.50 and 0.31 g cm^{-3} for monoUiO-66, monoUiO-66-NH2, and monoZr-Fumarate. Particularly noteworthy is the scale-up synthesis of the monoZrMOFs to a multigram scale, facilitating the processing of 1 L of Zr-Fumarate colloid to yield 67 g of monoZr-Fumarate. This scaled-up synthesis yields self-assembled monoliths while retaining all porosity and crystalline properties, signifying a notable development towards the practical industrial implementation of monolithic MOFs.158

On carbon capture, Fan *et al.* reported a γ -cyclodextrin-based monolithic MOF, mono γ -CD-MOF(K).¹⁵⁹ This monolithic material demonstrated high performance in terms of volumetric CO₂ uptake, achieving values of 44.04 and 36.68 cm³ (STP) cm⁻³ at 1 bar and 273 and 298 K, respectively, an improvement compared to the CO₂ uptake reported values of the powdered counterpart: 37.23 and 30.65 cm³ (STP) cm⁻³ at 1 bar and 273 and 298 K, respectively. The selectivity of CO₂ over N₂ was of 36.5, and it

displayed water stability after exposure to a 60% RH environment for 14 days. Breakthrough gas separation experiments using a 15/85 v/v CO₂/N₂ mixture—representative of post-combustion carbon capture-under both dry and 74% RH conditions showed that monoHKUST-1, monoUiO-66, and monoUiO-66-NH2 exhibited volumetric CO₂ uptake values of 22.6, 16.0, and 20.0 cm³ cm⁻³, respectively.¹⁶⁰ These values were significantly higher than their powdered counterparts, which reached only 12.4, 10.0, and 11.6 cm³ cm⁻³ in dry conditions.¹⁶⁰ Even in humid environments, the monolithic materials maintained nearly double the CO₂ uptake compared to their powdered forms. Similarly, for 50/50 v/v CO2/CH4 separation, relevant to natural gas purification, the monoHKUST-1, monoUiO-66, and monoUiO-66-NH2 monoliths exhibited superior volumetric adsorption performance, reaching 56.5, 42.0, and 36.2 cm³ cm⁻³, respectively. These values were significantly higher than those of their powdered analogues, which showed 30.8, 24.6, and 25.5 cm³ cm⁻³.¹⁶⁰ The monoliths maintained consistent performance even after undergoing five adsorption/desorption cycles, highlighting their robustness and stability.¹⁶⁰ Moving to COFs, we reported a self-shaped monolith using 1,3,5-tris(4-aminophenyl)benzene(TPB) and 2,5-dimethoxyterephthal-aldehyde (DMTP) as building blocks (Fig. 14).154 The COF monoliths were synthesized using a mixture of dioxane and acetonitrile. Using an acetonitrile volume fraction of 0.75 (v/v), the particle size was reduced to 40 nm, a size similar to what has been previously reported as necessary for achieving MOF monoliths (Fig. 14c).¹⁵⁴ The gravimetric BET surface area of this material, 2125 $m^2 g^{-1}$, is slightly lower but comparable to the values reported for non-self-shaped materials of similar COFs. For example, the BET surface area of 2535 $m^2 g^{-1}$ was reported for TAPB-PDA-AG COF aerogels by Illán et al. 161 However, in terms of volumetric BET area, the self-shaped monoliths surpass



Fig. 14 A self-shaped monolithic COF using TPB and DMTP as building blocks for gas separation. (a) Processing workflow for TPB-DMTP-COF monolith formation. (b) An optical image of TPB-DMTP-COF monolith. (c) Pore structure of TPB-DMTP-COF (C atoms are in grey, N atoms are in blue, and O atoms are in red; H atoms have been omitted for clarity). (d) SEM images of the TPB-DMTP-COF monolith synthesized with varying acetonitrile fractions (inset) – scale bar of 1 μ m. Adapted (reprinted) under the terms of the CC-BY 4.0¹⁵⁴ Copyright 2022, Elsevier.

the aerogel, with 332 vs. 43 m² (STP) cm⁻³. Dynamic breakthrough studies done with the obtained COF monoliths and using mixed gas feeds showed very similar separation performance for a 15% CO₂ – 85% N₂ mixture, and a noticeably sharper separation for the 50% CO₂ – 50% CH₄ mixture. For both cases, we observed an increase in CO₂ capacity (13.4% and 8.6%, respectively).

Furthermore, monolithic synthesis provides a pathway for structuring composite materials as well. Within the subfamily of MOFs, ZIFs especially stand out for their zeolitic topologies. ZIF-8, in particular, is a flexible MOF¹⁶² extensively studied for its ability to grow around smaller molecules, essentially acting as a scaffold, and encapsulating guest molecules. Mehta et al. reported the synthesis of a composite with SnO₂ NPs encapsulated using monoZIF-8.¹⁶³ To obtain the pristine MOF monolith, the reaction time was reduced to 15 minutes, maintaining a particle size of 60-80 nm. In contrast, for SnO₂@monoZIF-8, SnO₂ NPs were added to the 2-methylimidazole solution before mixing. This method yielded particle sizes ranging between 100 and 150 nm, suggesting that ZIF-8 nucleated at a slower rate in the presence of SnO₂ NPs. The composite exhibited moderate activity for the photocatalytic degradation of methylene blue (MB), achieving an average degradation of 41.5% with 53 µmol SnO₂-NPs loading. Subsequently, the degradation increased to 97.6% with a catalytic loading of 225 µmol. However, aggregation of SnO₂ NPs indicated their stabilization in the interparticle space rather than within the pores of the MOF, limiting their dispersion and loading capacity.¹⁶³ Ye et al. encapsulated sulforhodamine 640 (SRh) within monoZIF-8 - SRh, noted for its high photoluminescence quantum yield of around 63.6%, is an excellent candidate for laser gain mediums.164 The transparent ZIF-8 framework serves here as an effective scaffold for the dye, enhancing its optical performance. When excited with 532 nm

laser pulses (the second harmonic of an Nd laser), SRh@ZIF-8 displayed a sharp emission peak near 620 nm at an energy density of 31 μ J cm⁻². This represented the lowest threshold reported for SRh-doped polymers and MOF-based gain media at the time, surpassing SRh-doped polymers, which typically require 53 to 95 μ J cm⁻², and other MOF-based media with thresholds ranging from 41 to $7.5 \times 10^6 \mu$ J cm⁻². The observed narrowing of the emission band with increasing energy density indicated amplified spontaneous emission (ASE), reflecting the high optical quality of SRh@ZIF-8. With dimensions of approximately $6 \times 3 \times 1 \text{ mm}^3$, the composite is up to 10 000 times larger than conventional MOF crystals used in laser tests, demonstrating its potential for large-scale applications in laser devices and photonic technologies.

Tian et al. successfully encapsulated gold nanoparticles (Au NPs) within monoZIF-67 by reducing the reaction time to 30 minutes and conducting the reaction at 0 °C, resulting in Au NP sizes of approximately 140 nm, which is significantly smaller than the 260 nm particles obtained from earlier room temperature methods.¹⁶⁵ SEM imaging confirmed the uniform distribution and effective encapsulation of Au NPs within the ZIF-67 framework. The synthesis of $_{mono}$ ZIF-67 at 0 $^{\circ}$ C using a sol-gel method led to a structure with a higher volumetric BET area and CO₂ adsorption capacity compared to ZIF-67 powder, which in turn resulted in a 90% increase in the volumetric CO production rate. 8 mLAu@ZIF-67, the best performer, exhibited a volumetric CO production rate 1.5 times greater than monoZIF-67 and 3 times greater than ZIF-67 powder, reflecting a beneficial combination of high gravimetric yield and bulk density. Mechanical testing showed that 8 mLAu@ZIF-67 had an elastic modulus of 3.53 ± 0.25 GPa and hardness of 0.316 ± 0.040 GPa, which are similar to $_{\rm mono}$ ZIF-67's values of 3.47 \pm 0.15 GPa and

This article is licensed under a Creative Commons Attribution 3.0 Unported Licence.

Open Access Article. Published on 08 April 2025. Downloaded on 7/14/2025 10:04:51 AM.

 0.341 ± 0.027 GPa, indicating that the presence of Au NPs had minimal effect on mechanical stability. The successful synthesis of Au@ZIF-67 demonstrated improved CO₂ photoreduction performance, highlighting its potential for applications requiring both high performance and durability. 165

4.1.2. Dip coating. Dip-coating is a simple and widely adopted solution-based technique for depositing thin films and coatings on various substrates, ranging from metallics and ceramics to polymer films and fibers. The process involves immersing the substrate into a solution of the coating material, ensuring complete infiltration, followed by withdrawing the substrate from the solution. Subsequently, the wet coating sediments are evaporated to achieve dryness.¹⁶⁶ Mori and colleagues developed a strategy of submerging Cu-based reactors into an acidic solution containing an organic linker facilitating the in situ synthesis of HKUST-1 crystals on the reactor surface.¹⁶⁷ The formation of the MOF layer not only served as a catalyst for the deposition of active metal NPs but also led to the formation of carbonaceous layers through pyrolysis under inert conditions. This facilitated further functionalization with organic modifiers, e.g., p-phenylenediamine, and metal NPs, e.g., catalytically active Pd NPs, resulting in 3Dprinted reactors that showed promise for catalytic hydrogen production from liquid-phase hydrogen carriers. They then employed laser powder bed fusion (LPBF) to craft Cu-based reactors tailored for catalytic functions. LPBF, an additive manufacturing technique utilizing a high-power laser to selectively meld metallic powder layers, allowed for the creation of intricate three-dimensional structures with exceptional precision. By submerging Cu-based reactors into an acidic solution containing an organic linker, they facilitated the on-site synthesis of HKUST-1 crystals directly onto the reactor surface, enabling the fabrication of reactors with customized geometries and internal architectures, thereby enhancing their efficacy for catalytic applications. The resulting internal structure significantly influenced catalytic activity, as evidenced by the observed inverse correlation between pressure loss magnitude and pore size. Through LPBF-based adjustments in cell density and internal geometry, they could fine-tune reactors for specific catalytic processes, such as hydrogen generation from liquid-phase carriers.¹⁶⁷ Gkaniatsou et al. employed dip coating to deposit MOFs onto dehumidification heat exchangers using silicon as the binder, showcasing successful implementations.¹⁶⁸ They highlighted the use of water-stable Al-MOFs in adsorption cooling, synthesized through ecofriendly processes. MIL-160, CAU-10, Al-Fum, and CAU-23 exhibited potential, achieving thermal efficiencies above 0.6 and specific cooling powers exceeding 1 kW kg⁻¹. When powered by solar thermal energy, Al-MOFs maintained stable energy conversion efficiencies despite varying conditions. This approach ensured stable adherence, even through multiple temperature swing cycles. Dip coating ensured uniform coatings for optimal performance, although challenges with coating homogeneity arose at scale. The Al-MOF adsorption chiller showed promise for sustainable cooling, with future research aiming to enhance efficiencies and reduce costs for

broader adoption alongside increasing renewable energy integration.¹⁶⁸

In another work, Sarango et al. developed a dip-coating method for creating thin film nanocomposite (TFN) membranes with precise MOF (ZIF-8 and ZIF-67) nanoparticle arrangements.¹⁶⁹ This method reduces clumping and conserves reactants. For example, using ZIF-8 particles of 70 \pm 10 nm provided better uniformity and coverage than ZIF-67 particles of 240 \pm 40 nm, resulting in improved performance. The dip-coating process ensured MOF particle deposition without loss during interfacial polymerization. TFN ZIF-8 membranes showed increased methanol permeance (up to 8.7 L m² h⁻¹ bar⁻¹; a 150% rise compared to thin film composite membranes) while maintaining high rejection rates, indicating effective substance blocking and selective permeation.¹⁶⁹ COF membranes can also be fabricated through a dip-coating process. For example, Tsuru and coworkers created COF-1 nanosheets obtained by sonication of bulk COF-1 materials.¹⁷⁰ These nanosheets were then deposited onto the external surface of SiO2-ZrO2-modified α-Al2O3 supports using a drop-coating method, followed by drying at room temperature. The COF-1 membranes, about 100 nm thick, demonstrated high hydrogen gas permeability, reaching 17 mol m⁻² Pa⁻¹ s⁻¹ at 25 °C due to the perforations in the nanosheets. N2 adsorption isotherms showed that pristine COF-1 had a pore size of approximately 0.6 nm, with uniform 1.5 nm perforations in staggered nanosheet stacking. Moreover, the COF-1 nanosheets exhibited good thermal stability due to robust covalent bonds within the structure.¹⁷⁰

Apart from these top-down strategies, some reports about COF membranes are based on a bottom-up approach. For example, Park *et al.*¹⁷¹ deposited nine different COFs built using the same aldehyde precursor, 1,3,5-triphloroglucinol, on the surface of Zn electrodes by immersing a Zn foil into the COF precursor's solution. Then, they used an imine condensation reaction on the Zn surface to create uniform COF films of 30×12 cm² on both planar and curvilinear supports. The COF films showed strong affinity to Zn²⁺ ions due to favorable interactions with the electron-rich ketone and imine functional groups in the COFs, allowing for efficient mass and charge transport, and suppressing large Zn dendrites. The COF films did not show any noticeable cracks and deterioration after 200 folding/recovery cycles.¹⁷¹

Dip coating has also been used in sensing applications. Demessence *et al.*¹⁷² synthesized stable, 22 ± 5 nm monodisperse MIL-101(Cr) NPs using a green microwave method at 200 °C for 1 minute.¹⁷² The NPs, when dispersed in ethanol, can be stored for up to 2 months without structural changes. The NPs exhibit a Langmuir surface area of $4200 \pm 80 \text{ m}^2 \text{ g}^{-1}$, which is consistent with the bulk material. However, it is important to note that while the Langmuir model provides an estimation of surface area, it is not the most appropriate method for characterizing porous materials due to its inherent assumptions regarding monolayer adsorption. Thin films of MIL-101(Cr) deposited *via* dip-coating produce uniform films, the thickness of which depends on the concentration of the NP

suspension. The films demonstrate two-step water adsorption, corresponding to the two mesoporous cage sizes in MIL-101(Cr), with porosity reaching 78%, encompassing both the NPs and inter-NP space. Mechanical testing reveals the rigid nature of the NPs (Young's modulus $\approx 17 \pm 10$ GPa), while the overall film exhibits lower rigidity ($\approx 40 \pm 10$ MPa) due to its high porosity. Adsorption isotherms with alcohols showed reversible adsorption, indicating potential for selective adsorption properties.¹⁷² Overall, immersion time, temperature, concentration, and viscosity of the targeted suspension, together with the substrate surface properties, play significant roles in determining the thickness, distribution, and morphology of the thin films. Control over these parameters is essential to optimize their performance. Given these advantages, however, dip-coating techniques face challenges in terms of scalability, reproducibility, and parameter optimization.

4.1.3. Deposition techniques. Vapor deposition is a broad category of thin film deposition methods that involve vaporizing a source material into a gaseous state, to then condense onto a substrate. Vapor deposition can be classified into physical and chemical vapor deposition (PVD and CVD, respectively). PVD involves transforming a solid material into vapor through a physical process and, therefore, occurs typically in a vacuum environment to minimize gas interactions and

improve film quality. However, since this method requires elevated temperatures, it is challenging for MOFs and other reticular materials due to their limited thermal stability, with limited reports.¹⁷³ On the other hand, CVD involves controlled chemical reactions of gaseous precursor molecules on a substrate surface, resulting in the formation of a solid film layer with controlled thickness. This is a more versatile processing technique due to its scalability and, in certain cases, its ability to occur without necessitating a vacuum environment.¹⁷⁴

Despite the limitations of PVD on reticular, porous materials, PVD can also be combined with CVD. For example, Han *et al.* prepared HKUST-1 films using this approach (Fig. 15a).¹⁷⁵ They first created a 1 nm film directly on a SiO₂/Si(100) substrate using PVD, which served as the support for the subsequent deposition of H₃BTC *via* CVD at 200 °C. Glancing-angle X-ray diffraction (GAXRD) showed two distinct planes at (220) and (222), indicating the formation of highly oriented HKUST-1 thin film. In addition, Fischer *et al.* developed a femtosecond pulsed-laser deposition (femto-PLD) technique for fabricating ZIF-8 thin films (Fig. 15b).¹⁷⁶ This approach extends the available film fabrication techniques for MOFs, effectively sidestepping challenges associated with decomposition or amorphization. In this study, they used PEG-400 as a stabilizing agent for the deposition of ZIF-8, with the PEG being



Fig. 15 (a) Using a layer-by-layer growth approach, H_3BTC was sequentially deposited on a SiO₂/Si(100) substrate using CVD, and Cu using PVD. The graph shows the thickness of the resulting HKUST-1 thin film as the cycles of the layer-by-layer growth progress. Top and lateral views of the HKUST-1 thin film after 10 cycles as measured by AFM are inset. Reprinted with permission from ref. 175 Copyright 2019 Royal Society of Chemistry. (b) Femto-PLD technique for the fabrication of ZIF-8 thin films. A schematic illustration of the steps used for the fabrication of the thin films. Reprinted with permission from ref. 176 Copyright 2017 American Chemical Society. (c) Visualising the thin films: (i) optical image of the thin film on a sapphire substrate, (ii) SEM image of the thin film (top view, scale bar: 4 μ m), (iii) SEM image of the thin film (top view, scale bar: 200 nm) – crystals showing ZIF-8 morphology have been false-coloured as blue, (iv) SEM image of the thin film (cross-sectional view, scale bar: 4 μ m). Reprinted with permission from ref. 176 Copyright 2017 American Chemical Society.

removed by washing with ethanol after the formation of the ZIF-8 films. This methodology yielded mesoporous ZIF-8 films constituted by nanoscale ZIF-8 crystals, as confirmed by SEM (Fig. 15c) and 77 K N_2 isotherms.

Mondloch *et al.*¹⁷⁷ reported the first application of ALD – a variant of CVD based on sequential, self-limiting reactions that facilitate thickness control at the angstrom level¹⁷⁶ - for incorporating single atom sites (Zn and Al) inside the structure of NU-1000. This approach, termed 'atomic layer deposition in MOFs' (AIM), allowed to enhance the catalytic performance of NU-1000 in Knoevenagel condensation reactions. NU-1000, synthesized via solvothermal reactions, demonstrated thermal stability up to 500 °C, mesoporous channels, and strategically positioned -OH groups required for the metalation. Diethylzinc (ZnEt₂) and trimethylaluminum (AlMe₃) were used as ALD precursors to achieve metalation, giving an average of 0.5 Zn (Zn-AIM) and 1.4 Al (Al-AIM) atoms per Zr atom. NU-1000 retained its crystallinity during the process, while the BET area decreased from 2230 m² g⁻¹ for pristine NU-1000 to 1580 and 1160 m² g⁻¹ for the Zn- and Al-doped NU-1000, respectively, due to the extra metals and the space taken from the porosity.¹⁷⁵ Following on ALD, Stassen et al. showed the preparation of ZIF-8 films through a two-step 'MOF-CVD' method: a metal oxide deposition step and a vapor-solid reaction step.¹⁷⁸ The initial deposition involved creating a ZnO layer via ALD followed by the introduction of 2-methylimidazole organic linker vapor into the reaction system via CVD. Notably, the solvent-free nature of the MOF-CVD process allowed for liftoff patterning and the fabrication of MOF films on delicate substrates. In a separate study, Liu and coworkers used vapor-induced conversion within CVD to synthesize a series of large-area COF films featuring -C=N- linkages, termed PyTTA-TPA, PyTTA-BPyDCA, and PyTTA-BPDA, with controllable thicknesses.¹⁷⁹ Among them, the carrier mobility in a 30-nm-thick PvTTA-TPA COF film reached $1.89 \times 10^{-3} \text{ cm}^2 \text{ V}^{-1} \text{ s}^{-1}$, significantly higher than that of PyTTA precursors. The authors attributed this enhancement to charge transport through the COF lattice. Additionally, the film demonstrated notable electrocatalytic activity for the hydrogen evolution reaction (HER), outperforming metal-free COFs and even certain metallic catalysts.

4.1.4. Spray drying. Spray-drying involves the rapid atomization of a solution or suspension of multi-components of the desired materials into an aerosol droplet, followed by rapid solvent evaporation under hot air at a certain temperature and pressure, leading to the formation of a solid powder, normally with a spherical morphology.¹⁸⁰ During this process, several key parameters need to be controlled and optimized, including the feed rate at which the precursors are injected, the flow rate at which the atomization occurs, and the temperature of the gas that is used to dry the droplets formed. These parameters impact both at a process level (*e.g.*, throughput, cost, and quality) and the material itself (*e.g.*, size, shape, and morphology). Spray drying offers continuous manufacturing and maximizes throughput, ensuring consistency and efficiency throughout production.¹⁸⁰

Pioneering work in this area has been carried out by Maspoch and co-workers. First, Carné-Sánchez et al. introduced spray-drying for MOF synthesis (Fig. 16).¹⁸¹ They focused on the formation and subsequent drying of droplets containing the metal salt and the corresponding organic linker. Upon the rapid evaporation of the droplets, MOF nanocrystals were formed at the air-liquid interface. These nanocrystals merge to form hollow superstructures, whose size can be precisely controlled based on the type of MOF and the synthesis conditions, ranging from tens to hundreds of nanometers. The first example was using HKUST-1 by directly injecting a solution of Cu(NO₃)₂·2.5H₂O and H₃BTC in mixed solvents of DMF, EtOH, and H_2O (Fig. 16a). This approach was later expanded to other MOFs, including Cu-bdc, NOTT-100, MIL-88A, MOF-14, MOF-74, and UiO-66 (Fig. 16b-h).¹⁸¹ In addition to creating hollow structures, spray drying can generate dense ones. In this regard, Mitsuka et al. developed a two-step synthesis process for MOFs, focusing on those with high-nuclearity secondary building units-clusters containing multiple metal atoms that enhance stability and connectivity.¹⁸² The UiO-66 family exemplifies this, featuring Zr₆-based clusters that provide exceptional structural robustness. First, MOF seeds were generated by heating mixed precursors at a specific temperature. The seed suspension was spray-dried to promote crystal growth, resulting in spherical UiO-66 superstructures with diameters ranging from half a micron to a few microns, while the primary MOF particles typically remained below 100 nm in size. The process also allowed for control over particle morphology while facilitating MOF production.¹⁸¹ Garzon-Tovar *et al.* also developed a synthesis method for high-nuclearity MOFs by combining continuous flow and spray-drying synthesis.¹⁸³ This dual approach comparable to the work of Mitsuka et al. - was designed to produce spherical microbeads of MOFs, including UiO-66. Here, the continuous flow reactor ensured MOF nucleation and avoided the formation of amorphous products, which was an issue in earlier studies.^{181,182} The optimized process parameters included a feed rate of 2.4 mL min⁻¹, a flow rate of 336 mL min⁻¹, and an inlet temperature of 180 °C – these three parameters, and their impact on the quality of the materials through their BET areas, were largely explored. These conditions allowed complete solvent evaporation and resulted in spherical microbeads with an average diameter of 4.3 \pm 2.6 mm, composed of nanoparticle aggregates. The resulting UiO-66 beads exhibited a BET area of 1106 $m^2 g^{-1}$, similar to the one obtained using conventional methods. The method was also expanded to other high-nuclearity MOFs, including Fe-BTC/MIL-100 and $[Ni_8(OH)_4(H_2O)_2(L)_6]_n$. Fe-BTC/MIL-100 yielded 78% with a BET area of 1039 m² g⁻¹, while [Ni₈(OH)₄- $(H_2O)_2(L)_6]_n$ showed a yield of 60% and a BET area of 377 m² g⁻¹. Additionally, the method demonstrated flexibility in producing multivariate (MTV) MOFs, such as UiO-66, using different organic linkers.183

Camur *et al.* advanced the combined continuous-flow and spray-drying method on UiO-66-NH₂ and explored the effect of acetic acid as a modulator and using water as a solvent.¹⁸⁴ At 14% acetic acid, the microbeads exhibited a BET surface area of



Fig. 16 MOFs synthesized by the spray-drying approach (left: crystal structure, right: SEM images of the MOF superstructures and discrete nano-MOF crystals (inset)). (a) HKUST-1 (b) Cu-bdc. (c) NOTT-100. (d) MIL-88A. (e) MIL-88B. (f) MOF-14. (g) Zn-MOF-74. (h) UiO-66. Adapted with permission from ref. 181. Copyright 2013 Springer Nature.

840 m² g⁻¹, which increased to 1036 m² g⁻¹ at 56%. However, at 70% acetic acid, the BET area decreased to 655 m² g⁻¹ due to competition between the modulator and ligand, affecting crystallinity. The optimal concentration of 30% acetic acid resulted in microbeads with a particle size distribution of 4–10 μm, a BET surface area of 1261 m² g⁻¹, and a water uptake of 0.57 g g⁻¹ at 0.2 *P*/*P*₀. They expanded the method to Zr-fumarate, obtaining beads with a BET area of 664 m² g⁻¹ at 30% acetic acid.¹⁸⁵ Though slightly lower than hydrothermal methods, the spray-drying technique demonstrated scalability, ease of use, and environmental benefits, emphasizing the importance of modulator concentration in tuning MOF properties.¹⁸⁴ Boix *et al.* improved the integration of inorganic nanoparticles (iNPs) into 1.5 μm UiO-66 microbeads using a

flow reactor at 115 °C.¹⁸⁶ The UiO-66 and CeO₂@UiO-66 microbeads, along with their thiol-functionalized derivatives (UiO-66-(SH)₂ and CeO₂@UiO-66-(SH)₂), were synthesized using a continuous-flow spray-drying technique, forming spherical microbeads (average size: $1.5 \pm 1.0 \ \mu\text{m}$) composed of UiO-66 nanocrystals and CeO₂ nanoparticles. These microbeads showed high porosity, with BET surface areas of 945 m² g⁻¹ for UiO-66, 597 m² g⁻¹ for UiO-66-(SH)₂, 747 m² g⁻¹ for CeO₂@UiO-66, and 539 m² g⁻¹ for CeO₂@UiO-66-(SH)₂. They effectively removed heavy metals such as As(π and v), Cd(π), Cr(π and vi), Cu(π), Pb(π), and Hg(π) from a solution with 100 ppb concentration (for each metal ion precursor), with removal efficiencies of 99% for Pb(π) and Cu(π), 98% for Hg(π), 93% for Cr(π and vi), and 56% for As(π and v). Thiol-functionalization

enhanced adsorption capacities for $Pb(\pi)$, $Cu(\pi)$, and $Cr(\pi)$ and vi), while CeO₂ improved the removal of As(III) and Cr(IV). The microbeads exhibited stability during adsorption, with no detectable release of Zr(IV) or Ce(IV) ions and retained crystallinity after metal adsorption. In a continuous-flow column, CeO₂(a)UiO-66-(SH)₂ microbeads removed 99% of Pb(II) and Hg(II), 85% of Cd(II), 84% of Cr(III and VI), and 69% of As(III and v) at a flow rate of 1.3 mL min⁻¹, with a breakthrough time of 231 minutes (300.6 mL) and a maximum Cr(m) loading capacity of 82.7 mg g⁻¹. The microbeads were regenerated easily with an acidic treatment, achieving desorption rates exceeding 96%. In real river-water samples from the Buringanga, Bone, and Sarno Rivers, the microbeads reduced metal concentrations below WHO limits, demonstrating their effectiveness in real-world water purification. A magnetic version of CeO₂@UiO-66-(SH)₂ microbeads, incorporating Fe₃O₄ nanoparticles, allowed easy recovery from water using a magnet, maintaining their metal-adsorption capacity. The microbeads maintained integrity over three cycles, achieving a maximum capacity of 82.7 mg (Cr(m)) g⁻¹, and the Fe₃O₄-enhanced version enabled efficient recovery without compromising performance, further broadening their potential applications in water treatment. Continuing their work, they incorporated CeO2doped UiO-66 microbeads into porous polyethersulfone (PES) structures via spray-drying,¹⁸⁷ achieving CeO₂ encapsulation levels of 4.0% and 3.3%, with yields of 93% and 87%. Nitrogen adsorption confirmed a BET area of 945 $m^2 g^{-1}$ for UiO-66. In continuous-flow tests, 10 mg of microbeads removed over 99% of $Pb(\pi)$, $Hg(\pi)$, and $Cu(\pi)$ from 30 mL of water. Tests in river water from Bangladesh, Indonesia, and Italy showed over 98% removal of $Cd(\pi)$, $Cu(\pi)$, and $Pb(\pi)$, with $Cr(v_1)$ and $As(\pi)$ reduced to safe levels. The magnetically functionalized CeO₂/ Fe₃O₄@UiO-66-(SH)₂ microbeads demonstrated excellent performance and easy recovery, making them suitable for largescale water purification applications.

With some modifications, spray drying can be also translated to COFs. The challenge here is that COFs are typically formed under thermodynamic control while the spray-drying approach hinges on kinetic control - i.e. rapid product formation - which may not necessarily be thermodynamically stable and may result in different phases, including non-porous ones.¹⁸⁰ Garzon-Tovar et al.¹⁸⁸ introduced a method combining spray-drying with dynamic covalent chemistry to synthesize zero-dimensional spherical COF superstructures from iminebased nanocrystals. This two-step approach first forms amorphous polymer spheres via spray-drying, which are then crystallized into COFs like COF-TAPB-BTCA, COF-LZU1, and COF-TAPB-PDA. The resulting microspherical superstructures retain their size and shape after crystallization, with COF-TAPB-BTCA showing a BET surface area of 911 $m^2~g^{-1},~\text{COF-LZU1}$ at 319 m² g⁻¹, and COF-TAPB-PDA at 1162 m² g⁻¹. Furthermore, the method allows for the integration of functional materials, creating composites such as Rose-bengal@COF-TAPB-BTCA with uniform dye distribution and slow release, and Fe₃O₄@ COF-TAPB-BTCA composites that exhibited magnetic properties with 2.8% Fe₃O₄ content and easy magnet retrieval. This

approach expands the potential of COFs for applications requiring structured materials with enhanced properties.

4.2. Post-synthetic shaping and densification

While Section 3 describes the main differences and characteristics of conformed and densified bodies, the current section reviews the methods used to prepare them. Post-synthetic shaping provides an important degree of flexibility in controlling the final shape and body density by enabling control over compression conditions.¹¹⁷ To ensure optimal mechanical properties, these bodies require the introduction of binders and plasticizers.¹⁸⁹ The primary function of the binder is to glue the MOF powder particles together, while the plasticizer serves to increase the plasticity of the binder, making it less brittle. Binder materials can be either inorganic, such as Al₂O₃ or silica, or organic, such as mono/polysaccharides (e.g., cellulose) or polyvinyl alcohol (PVA); plasticizers include glycerol, propylene glycol, or triethyl citrate. Proper selection of the binder is crucial to the properties and performance of the resulting bodies to withstand crushing by attrition and heavy weight loads. Considerations such as surface tension and viscosity of the solution, as well as binder interactions with powder particles, are essential.¹⁹⁰ These considerations help strike a balance between preserving the desired material properties in terms of pore volume, density, and adsorption capacity and achieving the necessary mechanical properties and thermal conductivity in the final product.

Post-synthetic shaping can work differently when dealing with wet or dry samples. Solvents are often used to enhance binding between powder particles and binders, being removed during the drying, but they can also help to avoid pore collapse during the process.¹⁹¹ Wet processes takes place in three steps (Fig. 17a): (i) wetting and nucleation, when a volatile solvent and a binder are added and the mixture and the particles begin to aggregate; (ii) consolidation and coalescence, where the material starts aggregating until they reach a maximum size; and (iii) attrition and breakage, where the forces applied break and shape the material resulting in the final granules. In the cases where the material is sensitive to either the solvent or the heat required for drying - which can be the case in reticular materials - an alternative dry process that relies solely on mechanical compression may be employed.¹⁹⁰ Overall, one of the main drawbacks of this shaping process is the need for a precise control over the total pressure applied, the increase rate, and the dwell time in the mold during the pelletization. The process will allow for the increase in the density of the material and, therefore, an improvement in its mechanical properties (i.e. higher attrition or breakage resistance) and volumetric adsorption properties. However, as explained in Section 3, the increase in the density can result in a gravimetric reduction of the porosity due to pore collapse due to mechanical compression; alternatively, binder addition can result in pore blocking.¹³¹ It is important to note that, beyond certain mechanical pressures, which depend on each specific MOF, the crystalline structure transforms into an amorphous phase and the collapse of the porosity (Fig. 17b). Here, it is important to



Fig. 17 (a) The three steps for wet pelletization: (i) wetting and nucleation – when a volatile solvent (and if needed, a binder), is added and the mixture and the particles begin to aggregate, (ii) consolidation and coalescence – where the particles keep aggregating until particle size reaches its maximum, and (iii) attrition and breakage – where forces applied break and shape the material resulting in the final granules. (b) Partial amorphization of materials upon pelletization with pressure – leading to a reduction in the porosity of the material. As a consequence, depending on the application, values for these pelletization parameters need to be optimized.

remember that the mechanical properties of MOFs³⁶ and other reticular materials¹⁵⁴ depend first on topology and then on pore volume, pore size, and density *e.g.*, the smaller the cluster connectivity and the larger the pore volume and size, the lower the mechanical properties. Post-synthetic shaping and densification processes can be classified broadly into (i) tableting, (ii) extrusion, (iii) spherenoization, (iv) 3D printing, (v) phase inversion and hydrogelation, and (vi) glass formation.

4.2.1. Tableting. Tableting is a process where porous materials are shaped into solid forms (*e.g.*, pellets and granules) by compressing powder particles in a mold under high mechanical pressure, often resulting in a denser structure while maintaining the material's chemical integrity and porosity. This method is commonly used in the production of tablets for pharmaceutical and material applications. Ardelean *et al.* compressed MIL-101 MOF powder into pellets, achieving a peak bulk density of 1.34 g cm⁻³ at 120 MPa, where the transition to an amorphous phase starts.¹⁹² However, at intermediate densities up to 0.47 g cm⁻³ (30–50 MPa), the crystal structure remained intact, and pellets within the 0.45 to 0.47 g cm⁻³ range exhibited a hydrogen storage capacity of 40 g L⁻¹ at 196 °C and 8 MPa. At 77.3 K, hydrogen adsorption isotherms

showed reduced surface area and micropore volume as density approached crystal density, while XRD indicated no significant structural changes until then. IR spectra revealed pressureinduced shifts in carboxylate and phenylene frequencies, paralleling findings in MOF-5 and MOF-177, which transitioned to amorphous phases at higher densities.¹⁹² Oh et al.'s research measured the hydrogen uptake capacity of MIL-101 in powder and pellet forms, showing that at 20 K, pellets can absorb up to 9.6 wt% and 42 g L⁻¹ of hydrogen.¹⁹³ This cryo-adsorption method offers a wider temperature range for storage without boiloff, making it ideal for industrial use. The hydrogen capacity at 20 K is nearly double that at 77 K due to pore condensation in MIL-101's cavities. The study suggests that combining cryo-adsorption with liquefaction at 20 K can reduce boiloff and extend the storage range to 37 K, with MIL-101 sorbents preventing rapid hydrogen expansion and improving system performance.¹⁹³ Then, Blanita *et al.* developed hexagonal prism-shaped MIL-101 monoliths for hydrogen adsorption, achieving envelope densities up to 0.467 g cm⁻³ and good mechanical stability.¹⁹⁴ At 77 K, excess hydrogen adsorption $(N_{\rm ex})$ decreased from 5.69% H₂ to 4.54% H₂ for pellets with a density of 0.467 g cm⁻³, and dropped to 2.4% H₂ at 159 K. At

View Article Online Review Article

77 K and 150 bar, the total volumetric capacity was 46.5 g L⁻¹, with a working capacity of 45 g L⁻¹ after discharge at 159 K and 5 bar. MIL-101 pellets (0.4 g cm⁻³) stored 6.9 kg H₂ at 100 bar and 7.9 kg H₂ at 150 bar, corresponding to volumetric capacities of 36.4 g L⁻¹ (6.2% H₂) at 100 bar and 41.1 g L⁻¹ (7.0% H₂) at 150 bar. MIL-101(Cr) remained stable in air for 8 months, withstood temperatures up to 220 °C, and supported over 1500 cycles, making it suitable for low-purity hydrogen refuelling.

Tableting and pelletization can be applied beyond hydrogen storage. Permyakova et al. used MIL-127(Fe), MIL-125(Ti)-NH₂, MIL-100(Fe), and MIL-160(Al) for water adsorption in heat storage.¹⁹⁵ For example, they evaluated MIL-160(Al) granules in a pilot-scale reactor. Through a wet granulation process, the MOF powder was mixed with 10 wt% silica sol as a binder, and then shaped into spherical granules using a rolling machine. After drying at 100 °C for 12 hours, they obtained spherical macrostructures ranging from 0.5 to 1.8 mm in size. The BET area dropped from 1150 to 1000 $m^2 g^{-1}$ only, likely due to partial pore blockage caused by the binder. Despite these changes, MIL-160(Al) granules maintained good cycling loading lifts over 10 adsorption/desorption cycles, with water working capacities of 0.36 and 0.32 $g_{water} \; {g_{adsorbent}}^{-1}$ for powder and granules, respectively, measured between 30 °C for adsorption and 80 °C for desorption, at 1.25 kPa - corresponding to an energy capacity of 305 W h kg⁻¹ under mild desorption conditions. Similarly, Kim et al. used wet granulation for MIL-100(Fe) and a silica sol binder for SF6/N2 separation, obtaining granules with sizes ranging 1.18-1.70 mm.¹⁹⁶ The granulation process resulted in a slight reduction in BET area from 1772 m² g⁻¹ (powder) to 1619 m² g⁻¹. This is an 8.6% decrease in BET area, lower than the 33.6% reduction reported in pressed granules. In turn, the bulk density increased from 331 to 498 g L^{-1} , enhancing the volumetric adsorption capacity. At the end of the day, the SF₆ adsorption capacity of the granules (1.658 mmol g^{-1}) was similar to the powder (1.673 mmol g^{-1}), maintaining structural stability after high-temperature exposure and five adsorption/desorption cycles. Breakthrough experiments with 10 vol% SF₆/N₂ mixtures showed that MIL-100(Fe) granules had an SF₆ breakthrough time that increased linearly with pressure. Although Zeolite 13X performed better at lower pressures, MIL-100(Fe) demonstrated improved performance at higher pressures, with faster regeneration (20 minutes vs. 250 minutes for Zeolite 13X) and consistent performance across five cycles, while Zeolite 13X experienced some performance decline due to difficult desorption. In another study, Martins et al. used a method to create MIL-100(Fe) granules by mixing the MOF powder with 10% silica binder and spraying water and ethanol, resulting in semi-spherical granules of 1.0-3.0 mm diameter, with a micropore volume of 0.58 $\text{cm}^3 \text{ g}^{-1}$ and a BET area of 1568 m² g⁻¹.¹⁹⁷ Using pressure swing adsorption (PSA), they achieved 99.5% ethane purity (86.7% recovery) and 99.4% propane purity (97.0% recovery) in 30/70 ethane/propane mixtures, and 100% ethylene purity and 94.7% propane purity (100% recovery) for 30/70 ethylene/propane mixture.¹⁹⁷

When using pelletization to structure porous materials, several factors must be considered. One key issue is the

mechanical properties of highly porous materials. While pelletization can increase volumetric capacities for adsorption, it can also lead to a 15–20% decrease in gravimetric capacities compared to the powder form. This reduction is primarily due to the loss of capacity from the addition of binders.¹⁸⁹ While similar developments and densification techniques to the ones reported here have been widely used for materials such as zeolites, especially in applications such as chemical separation and conversion,¹⁹⁸ the application of these techniques to newer classes of reticular porous materials has not yet been thoroughly investigated. Given the open nature of their porosities, these emerging classes of porous materials present unique challenges, and further research may be needed to adapt and optimize densification methods for their specific properties and applications.

4.2.2. Extrusion. Extrusion is a well-established technique widely utilized to shape materials. Extrusion broadly involves the movement of material through confined spaces, typically facilitated by a piston or sets of screws. Standard extrusion setups comprise a feeder module, which operates through either volumetric or gravimetric methods, and a barrel housing either a piston screw (Fig. 18a), or two screws (Fig. 18b). In piston extrusion, hydraulic or pneumatic pressure is applied to materials within a die, enabling precise shaping of complex forms. On the other hand, in screw extrusion, the screw(s) convey the material through the interior of the barrel, subjecting it to shearing forces, before its eventual exit from the barrel. At the exit point, a die is used to shape the material according to the desired application. It is worth noting that similar setups are employed in mechanochemistry approaches, which are solvent-free strategies for the large-scale synthesis of materials.¹⁹⁹ While the mechanochemistry approach to synthesis should be theoretically classified under in situ shaping techniques, it is discussed here.

In the case of MOFs, the extrusion process typically involves a combination of MOF powder and, like in pelletization, a binder, and a plasticizer. Extrusion is also typically performed in the presence of a solvent, following stages that include pressurization and final shaping.²⁰⁰ While in classical porous materials such as zeolites, binders can be subsequently removed through heat treatment processes (typically in the range of 300 to 1000 °C)^{201,202} after shaping – resulting in the formation of macropores - this is more difficult in the case of MOFs due to the low thermal stability. Janiak et al. employed several hydrophilic, organic binders to shape commonly used hydrostable MOFs, incorporating a freeze-drying step post the extrusion process.²⁰³ Küsgens et al. prepared HKUST-1 monoliths by extruding a slurry mixture comprising the MOF, methyl hydroxypropyl cellulose, and methoxy-functionalized siloxane ether.²⁰⁴ This method exhibited a high MOF loading per monolith, presenting an almost pure structure, compared to a coated substrate monolith. They demonstrated enhanced structural robustness, withstanding forces of up to 320 N, surpassing cordierite monoliths deposited through in situ growth of HKUST-1. In another study, Tsalaporta et al. shaped four different MOFs - UiO-66, ZIF-67, HKUST-1, and ZIF-8 - into



Fig. 18 A schematic depiction of the extrusion process. A standard extrusion setup typically comprises a feeder module that operates *via* volumetric or gravimetric methods, and a barrel housing either piston screws, a screw or two screws. (a) A piston extrusion setup. (b) A screw extrusion setup. Reprinted with permission from ref. 119 Copyright 2021 Royal Society of Chemistry.

granules using methylcellulose and bentonite as binders.¹⁵⁵ ZIF-8 remained stable after the granulation process, whereas HKUST-1 and UiO-66 exhibited a reversible partial loss of crystalline morphology when pelletized with water, while ZIF-67's crystal structure was irreversibly lost.

Khabzina et al. produced UiO-66-COOH using a piston extruder, achieving an 89% yield and a space-time yield of 350 kg per day per m^3 in an aqueous batch reactor without using toxic chemicals or organic solvents.²⁰⁵ They tested both freeze-granulation and extrusion for NH₃ capture at 600-1200 ppm and relative humidity levels of 0%, 40%, and 70%. UiO-66-COOH pellets and extrudates achieved NH₃ uptakes of 55 and 53 mg g^{-1} , respectively, compared to 30 mg g^{-1} for Norit and 39 mg g^{-1} for 3 M commercial, activated carbons. The original MOF had a BET area of 710 m² g⁻¹, which reduced by $\sim 50\%$ after mechanical compression, with pellets and extrudates showing 359 and 418 m² g⁻¹, respectively. Bulk densities varied, with compressed powder at 0.62 cm³ g⁻¹, extrudates at 1.04 cm³ g⁻¹, and beads at 0.12 cm³ g⁻¹. Attrition tests showed less than 2% weight loss, and NH₃ uptake remained stable after 7 days of aging at 80% humidity (34 mg cm⁻³ vs. 33 mg cm⁻³). Regeneration at 150 °C retained 70–77% NH₃ capacity.

Hong *et al.* used a single-screw extruder with bentonite clay as a binder to create Cr-based MIL-101 monoliths for CO_2 adsorption.²⁰⁶ Monoliths were formed by mixing MIL-101(Cr) powder, bentonite clay, and water into a paste, which was extruded, dried first at 10 °C and then at 150 °C for 33 hours. The monoliths contained up to 75% (w/w) MIL-101 (Cr), were cut into 7 cm lengths, and had uniform channel sizes of 0.90 mm for consistent gas flow during dynamic adsorption tests. Characterization included PXRD to confirm crystal structure and SEM to reveal the cubical structure. Mercury intrusion porosimetry (MIP) indicated porosity values of 4.42% for the purified powder and 17.93% for the monoliths. Radial compression strength tests showed that monoliths with 60% and 75% weight MOF/binder ratios had elastic moduli of 10.60 N mm⁻² and 4.97 N mm⁻², respectively. The MIL-101(Cr) extrudates exhibited a small reduction in BET area, with 183 m² g⁻¹ compared to 202 m² g⁻¹ for the powder. However, when comparing the CO₂ adsorption capacities at 2 bar and 25 °C, it showed a more important decrease from the 1.44 mmol g^{-1} of the powder down to 0.91 mmol g^{-1} for the extrudates. In a follow-up study, Hong et al. compared the performance of honeycomb MIL-101(Cr) extrudates with zeolite 13X.¹¹⁰ They found that MIL-101(Cr) monoliths have 1.3 times higher porosity than 13X zeolite monoliths. Specifically, MIL-101(Cr) monoliths demonstrated better CO2 mass transfer, with breakthrough and equilibrium times reduced by approximately 20% and 35%, respectively, compared to 13X zeolite monoliths. At breakthrough, the CO₂ adsorption capacity of MIL-101(Cr) monoliths was about 37% higher (in mmol g^{-1}) than that of 13X zeolite monoliths, while at equilibrium, it was about 7% lower. Overall, MIL-101(Cr) monoliths showed 1.5 times greater efficiency for CO₂ adsorption than 13X zeolite monoliths. The study also found that higher regeneration temperatures enhanced CO₂ adsorption capacity for both types of adsorbents.

4.2.3. Spheronization. Spheronization, typically used after extrusion, is a widely employed method for producing spherical bodies. In this process, extruded structures are uniformly cut and transformed into spherical shapes through plastic deformation using a spheronizer.²⁰⁷ A typical spheronization apparatus comprises a hollow vertical cylinder and a horizontally rotating grooved 'friction' plate located inside. The cylindrical extrudates are dumped onto the spinning friction plate and are processed to form agglomerate shapes with nearly uniform diameters.²⁰⁸ Spherical structures typically manifest through one of two primary mechanisms. The first mechanism involves the gradual transformation of cylindrical structures, which initially exhibit sharp edges, into cylindrical shapes with rounded edges. Then, these structures evolve into dumbbelllike and elliptical particles before ultimately assuming spherical configurations. The second mechanism suggests that cylindrical structures experience torsional forces, leading to the creation of cylinders with rounded edges, which then fragment into discrete segments. Under the influence of

rotational and frictional forces, these individual segments transform to form spherical particles. Importantly, this process may yield spherical particles with distinctive surface features, such as characteristic grooves or cavities.²⁰⁸ The quality of the resultant spheres and their corresponding particle size distributions are influenced by several process parameters. Notably, factors such as load, duration of the process, and rotational speed of the spheronizer play pivotal roles. For instance, as the spheronizer speed increases and the load decreases, there is a reduction in yield. Conversely, increasing both the spheronization time and load increase the yield. Furthermore, particle sizes exhibit an inverse relationship with the rotational speed. The speed parameter exerts additional influence on critical attributes such as pellet hardness, porosity, and bulk (particle) density, thereby contributing to the overall quality of the pellets.207,208

For example, Dhainaut et al. employed extrusion-spheronization to shape two MOFs, UiO-66 and UiO-66-NH₂.²⁰⁹ They used biosourced chitosan and hydroxyethyl cellulose (HEC) as binders. They noted a preservation of the physicochemical properties of the initial powdered materials, with the BET area experiencing a reduction ranging from 5 to 33%, depending on the MOF and binder employed. Interestingly, there was a nonlinear decline in the BET area, ranging from 5% to 38% reduction in BET area for binder quantities ranging from 2.0% to 5.6% by weight. Importantly, the shaping process substantially enhanced the mechanical strength of the MOFs investigated, while preserving their efficacy in capturing iodine, krypton, and xenon.²⁰⁹ In a separate study, Ren et al. developed a method to shape UiO-66 powder into spherical pellets ranging from 0.5 to 15 mm in diameter, using 10 wt% sucrose as a binder, rather than relying on mechanical pressing. This process, which produced kilogram-scale batches in just 30 minutes through centrifugal granulation, showed promising results.¹²⁰ Durability tests showed no breakage after 70 consecutive drops from 0.5 m and only 5% breakage after 60 minutes of tumbling at 25 rpm. SEM images confirmed that 0.5-2 µm UiO-66 crystals were tightly bound by sucrose, maintaining interparticle space, which facilitated hydrogen diffusion, with BET areas of 674 and 1367 $m^2 g^{-1}$ for the spheres and powder, respectively. This decrease in porosity also reduced the hydrogen storage at 77 K and 1 bar from 1.54 wt%, powder, down to 0.85 wt% for the pellets.

4.2.4. 3D-printing. 3D printing – also referred to as additive manufacturing – is a manufacturing process that creates 3D objects from digital files. This technique operates on the principle of layer-by-layer additive construction, where materials are incrementally deposited and built upon each other to create the final object. Numerous 3D printing methods exist, including stereolithography, digital light processing, selective laser sintering, and binder jetting, among others.²¹⁰ This precision fabrication technology offers multiple advantages, such as unparalleled design flexibility, rapid production capabilities, and the ability to generate sophisticated geometries that would be exceedingly challenging to achieve with traditional manufacturing methods. Notably, 3D-printing has found extensive

applications in the biomedical field. For example, Hsieh et al. reported how polyurethane-gelatin hydrogels' mechanical properties augmented when adding ZIF-8 as a network enhancer.²¹¹ The addition of a small quantity of MOF $(\leq 750 \text{ mg mL}^{-1})$ notably enhanced modulus features, shearthinning behavior, and structural stability without compromising printing properties or water capacity. For the 3D printing, various amounts of 500-900 nm ZIF-8 crystals (50, 125, 1250, and 3750 mg mL⁻¹) were incorporated into a polyurethane/ gelatin bio-ink to improve modulus. An optimal MOF concentration of 875 mg mL⁻¹ was determined for cell survival and proliferation. The resulting 3D-printed MOF composite bioinks demonstrated excellent stackability and printability for constructing blood vessels and ear-shaped structures. Lim et al. used colloidal gels made of ethanol and HKUST-1 NPs as inks to craft 3D printed MOF monoliths (Fig. 19a).²¹² The 3D-printed HKUST-1 structure showed a BET area of 1134 m² g⁻¹ along with a 0.61 $\text{cm}^3 \text{g}^{-1}$ total pore volume. (Fig. 19b and c). Highpressure CH₄ adsorption at 65 bar and room temperature showed 271 and 131 cm³(STP) cm⁻³ for the 3D-printed HKUST-1 structure and powder, respectively (Fig. 19c). The former adsorption capacity is similar to the reported values of free standing, sol-gel HKUST-1 monoliths.¹⁸ The printed monolith showed a reasonable hardness of 42 Vickers Hardness (HV). Moreover, the structural integrity was deemed suitable for gas storage applications, offering promising avenues for various configurations including microreactors and adsorbent beds.

4.2.5. Phase inversion and hydrogelation. Phase inversion is a process for the transformation of a thermodynamically stable polymer solution from a fluidic state to a solidified state. This transformation is initiated by the liquid-liquid de-mixing phenomenon, where the initially homogeneous polymer solution segregates into two distinct phases. One phase becomes enriched with polymer molecules, forming a dense and concentrated region, while the other phase exhibits a lower concentration of polymers.²¹³ As this segregation process progresses, the polymer-rich phase undergoes solidification through various mechanisms, such as gelation or crystallization. These processes lead to the formation of a sturdy and continuous solid membrane structure within the material. Concurrently, the phase containing a lower concentration of polymers facilitates the creation of pores or voids within the material matrix. These pores contribute to the overall porosity and permeability of the resulting membrane. The interplay between the polymer-rich and polymer-lean phases during phase inversion ultimately yields materials with tailored morphological features optimized for applications of interest in the present context.²¹³ The de-mixing process can be induced by four common techniques: reducing the temperature, immersing the material in nonsolvent baths, eliminating volatile solvents from the solution, or precipitating from a vaporphase.^{213,214} Among these, one prevalent method involves immersion precipitation, where the polymer solution is immersed in nonsolvent baths. Nonsolvent baths are solutions that do not dissolve the polymer but instead induce de-mixing.



Fig. 19 3D-printed HKUST-1 monoliths boasts impressive BET surface areas of $1134 \text{ m}^2 \text{ g}^{-1}$ – with a substantial mesopore volume. (a) Optical images of different HKUST-1 3D printed structures. High-pressure methane adsorption tests at 90 bar and room temperature unveiled the exception capacity of the monolith. (b) Gravimetric and (c) volumetric absolute methane uptake isotherms for 3D-printed monolith and the powder forms. Adapted from ref. 212 Copyright 2019 American Chemical Society.

These baths typically consist of substances in which the polymer has low solubility or does not dissolve at all. For example, water can often act as a nonsolvent for many organic polymers. When the polymer solution comes into contact with the nonsolvent bath, it triggers de-mixing by altering the solventpolymer-nonsolvent interactions. This change in interactions leads to the formation of two distinct phases: one rich in the polymer and the other containing a lower concentration of polymer. The bath provides an environment conducive to the controlled separation of these phases. By carefully selecting the composition and conditions of the nonsolvent bath, it becomes possible to exercise control over the de-mixing process, allowing for the fabrication of membranes with desired morphological characteristics.²¹³ While there are several thermodynamic considerations based on ternary phase diagrams of polymer/ solvent/nonsolvent systems and kinetic factors related to mass transfer rates of solvent and nonsolvent, these concepts fall beyond the scope of this review. Readers interested in expanding on this topic are encouraged to refer to an excellent review by Holda and Vankelecom.213

Phase inversion has been used for the shaping of MOFs to exploit their catalytic properties for the neutralization of chemical warfare agents (CWAs). Peterson et al.²¹⁵ developed reactive, MOF-polymer composite beads, with a size ranging from 300 µm to 2 mm, using phase inverted poly(styrene-blockethylene-ran-butylene-block-styrene) (SEBS). Due to the bulky nature of the polystyrene (PS) blocks, there was a low infiltration of the polymer into the pores of the MOF, thereby preserving the core functionality while imparting viscoelasticity to the resulting composite material. The resulting composite showed better CWA removal capabilities and reactivity in comparison to activated carbon fabrics - which were the previous state-of-the-art. Along similar lines, Stylaniou et al.²¹⁶ developed spherical composite beads (MOF@polymer beads) from UiO-66-NH₂, UiO-66-pyridine, and UiO-67-(NH₂)₂ combined with poly(vinylidenefluoride) (PVDF), PS, and poly(ether sulfone) (PES).²¹⁷ Here, the MOF was first synthesized and then combined with the polymer. By doing so, it retains the crystallinity of the MOF and the accessibility to the pores. Composite beads formed from UiO-66-NH₂ and PES showed good catalytic performance in converting dimethyl *p*-nitrophenylphosphate (DMNP) to dimethyl phosphate (DMP), achieving a conversion rate of 62% in just 5 minutes. This catalytic efficiency may be attributed to the presence of interconnected macropores. Additionally, the composite beads's activity was kept over three cycles.

Beyond the phase-inversion method, there have been several attempts to shape MOFs as composite beads, such as using gelation-based techniques. For example, Valizadeh et al. synthesized a UiO-66 analogue using double amino functional groups in the linker (UiO-66(NH_2)₂) and then shaped it into MOF@PES beads for the removal of Cr(vi) from water.²¹⁸ The formation of composite beads addresses several challenges such as clogging, pressure drop, and material loss, which are often encountered when loading the powder into columns. The composite recorded a high $Cr(v_1)$ uptake of 135 mg g⁻¹, while being fully recyclable - tested in real-world samples. Additionally, the integrated process, which was performed in a glass column equipped with a visible light source, allowed for the photoreduction of Cr(vi) solution to less toxic Cr(iii) species during adsorbent regeneration, an interesting approach for Cr(vi) removal in a single continuous process.²¹⁸ Yang et al.²¹⁹ developed a polymerization strategy for the preparation of MOF-polymer composite beads using biocompatible and biodegradable poly(acrylic acid) (PAA) and sodium alginate monomers. The method involved the formation of double-crosslinked networks of PAA, sodium alginate and Ca+2 ions in water. The introduction of PAA made the beads highly stable due to hydrogen-bonding and ionic interactions - making them a promising strategy for liquid separations. The strategy allowed the formation of stable composite beads for 15 structurally diverse MOF systems (Fig. 20), MIL-101(Cr), MIL-100(Fe), HKUST-1, UiO-66, ZIF-8, ZIF-67, and MIL-100(Fe)/ PDA; the beads exhibited a Pd uptake of 498 mg g^{-1} . The method was applicable for large-scale structuring of the reticular porous materials using a continuous flow system driven by a peristaltic pump.²¹⁹ An alternative solution involves the utilization of MOF-cellulose composite beads.²²⁰ By



Fig. 20 A polymerization process using biocompatible and biodegradable monomers, cross-linked with calcium ions has been developed for the preparation of MOF–polymer composite beads. Optical images of beads of (a) MIL-127-Fe, (b) MIL-101(Cr), (c) HKUST-1, (d) UiO-66, (e) NH_2 -MIL-53(Al), (f) CuTDPAT, (g) ZIF-67, (h) ZIF-8, (i) Ni-pyrazolate, (j) Fe-BTC, (k) Fe-BTC/PDA, (l) Fe-BTC/PpPDA, (m) and (n) magnetic Fe₃O₄/MIL127-Fe, (o) Eu₂(BDC)₃, and (p) Tb₂(BDC)₃ as well as (q) Eu₂(BDC)₃ and (r) Tb₂(BDC)₃ under ultraviolet illumination with a wavelength of 254 nm. Adapted with permission from ref. 219 Copyright 2022, American Chemical Society.

embedding MOFs within biodegradable sodium CMC, MOFcellulose composite beads were created. The drying was done *via* two different methods of heat drying and freeze-drying. This structure maintains MOF crystallinity and porosity (with the drop in the BET area being less than 20% for both drying techniques compared with the original powder) while providing the composite beads with robust mechanical properties. To demonstrate practicality, they showed that MIL-100/CMC-HD composite beads effectively degrade more than 95% of dyes and are amenable to multiple cycles of reuse.

4.2.6. Glass formation. While MOFs have long captivated researchers with their porosity and crystalline structure, there has been a growing interest in exploring their amorphized phases, revealing new dimensions of versatility and potential applications. This area of research was initiated by pioneering work from Bennett, Goodwin and Cheetham, who first demonstrated the amorphization of ZIF-4, a prominent MOF variant.²²¹ Their study laid the foundation for deeper investigations into the behavior and properties of amorphized MOFs. Over the years, Bennett, Horike and others have expanded this work, systematically growing upon the early findings and coining the term 'MOF glasses.' Glass formation in materials like

coordination polymers (CPs) and MOFs can be achieved through various methods beyond traditional melt-cooling processes. Techniques such as melt-quenching, mechanical induction (namely via ball-milling), and direct glass synthesis have been successfully used to create consistent amorphous structures. Each of these techniques will be briefly introduced here. In a melt-quenching process, a crystalline MOF is heated to temperatures above its melting point in order to obtain a 'melt' state. The melting temperatures of MOFs can vary based on their specific composition and structure. Typically, MOFs exhibit relatively low melting points compared to conventional inorganic materials. Generally, the melting point of MOFs falls within the range of 350-750 °C. This melt state is then cooled rapidly - in a process called 'vitrification' - in order to obtain a 'glassy' state. This is only achievable when the melt state is stable, which in itself is not a usual phenomenon.¹⁴² Indeed, this phenomenon has been just observed for a few classes of families of CPs/MOF, including some phosphate-azole frameworks, ZIFs, thiocyanate and nitrile-based frameworks, and metal-bis(acetamide) frameworks. Alternately, a glassy state may be induced via the introduction of mechanical stimuli through processes such as ball-milling. Furthermore, it is

possible to directly synthesize amorphous MOFs exhibiting glassy behavior using methods analogous to the sol–gel method discussed in Section 4.1.1.¹⁴² For more detailed discussions on these aspects, we refer the reader to excellent reviews dedicated to glassy MOFs and the methods for their preparation.^{34,142}

In terms of the performance and application of MOF glasses, Wang et al. developed a glass-based membrane from the mixedlinker framework ZIF-62.¹⁴⁷ The framework was chosen for its glass-forming ability, as it undergoes a melting process prior to decomposition without interfering with its immediate recrystallization. While the ZIF-62 glass retained some porosity, there was a noticeable drop in the gas uptakes from 18.5 to 11 $\text{cm}^3 \text{g}^{-1}$ for the crystalline and the glass, respectively, for CO_2 at 1 bar and 293 K; 10 and 2.6 cm³ g⁻¹ for the crystalline and the glass, respectively, for CH₄ at 1 bar and 293 K; 2.4 and $0.7 \text{ cm}^3 \text{ g}^{-1}$ for the crystalline and the glass, respectively, for N₂ at 1 bar and 293 K. The permeance rates at room temperature for H₂, CO₂, N₂, and CH₄ were of 22, 9.7, 0.41, and 0.37×10^{-9} mol $m^{-2} s^{-1} Pa^{-1}$, respectively. They additionally fabricated a composite MOF glass membrane on porous ceramic alumina support using a melt-quenching approach. They, however, faced challenges with regards to an even spread of the melt on the support due to its high viscosity. Nonetheless, the resulting membranes were grain-boundary free, having the potential for long-term stability - they demonstrated no loss in permeance and selectivity for over two days. They exhibited excellent separation performance for H₂/CH₄, CO₂/N₂ and CO₂/ CH₄, with selectivities of 50.7, 34.5, and 36.6, respectively.¹⁴⁷ With regards to the somewhat rarer in situ glass formation, Yaghi and co-workers showed how a slow evaporation of a solution containing Ti-oxo clusters, fumaric acid, and m-cresol in a mixture of ethanol and tetrahydrofuran led to the formation of carboxylate linkages between the cluster and the linker, resulting in a transparent glass.²²² The glass - named Ti-Fum – had a record-high BET area of 923 $m^2 g^{-1}$, a value much higher than typical MOF glasses.²²²

5. Considerations for industrial translation

5.1. Influence of structuring methods on structural integrity under extreme conditions

The industrial implementation of porous reticular materials requires not only a high degree of control over their structuring but also a thorough understanding of how these methods influence their mechanical robustness, chemical stability, and long-term durability under extreme operational conditions. As discussed in previous sections, in industrial applications, these materials are often exposed to harsh environments that can lead to framework degradation, pore blockage or collapse, and loss of crystallinity.³³ Mechanical stability is particularly important for gas storage, separation, and catalysis, where materials experience high pressures during adsorption and desorption cycles. Structuring methods such as densification, extrusion, and sol–gel monolith formation may significantly impact the

ability of these materials to withstand mechanical stress. Densification techniques are commonly used to improve volumetric performance but can result in reduced porosity and pore connectivity.99 Recent studies have shown that hierarchical structuring, such as templated assembly and freeze-casting, can enhance mechanical stability by introducing reinforcing architectures that prevent pore collapse under pressure while maintaining high porosity.^{203,216} Additionally, hybrid structuring approaches combining sol-gel processing have demonstrated improved resistance while retaining surface area and functionality.96 For instance, the sol-gel processed monoHKUST-1 displayed a Young's modulus which matches its conventional powder counterpart, while having a 130% hardness due to its higher density.¹⁸ Another strategy to mitigate pressure-induced framework collapse involves incorporating secondary support structures or polymer binders into monolithic frameworks. Studies on polymer/MOF composites have reported improved mechanical stability while maintaining adsorption performance, particularly in gas storage applications.^{219,223} However, the main challenge here lies in optimizing binder selection to avoid pore blockage and loss of accessible surface area.

Conventional MOFs with weak metal-ligand bonds, such as Zn-based frameworks with carboxylic acid based linkers, start degrading above 300 °C, while higher-temperature-resistant MOFs like UiO-66 offer greater stability due to their stronger coordination bonds.^{224,225} However, for long-term industrial applications such as catalysis, membrane separation, and gas capture, MOFs should not operate above 150 °C to prevent structural degradation. The structuring method plays a crucial role in thermal resilience, as shaping techniques can either reinforce or compromise mechanical integrity. COFs, which rely on covalent rather than coordination bonds, generally exhibit higher thermal stability than MOFs.⁹ However, structured COF forms, including fibers, membranes, and aerogels, may become unstable under extreme heat and pressure.²²⁶ Strategies such as cross-linking and carbonization have been explored to enhance their durability while maintaining porosity. Pyrolyzed COF monoliths have shown excellent thermal stability in catalytic applications, making them promising for high-temperature environments.^{227,228} To ensure industrial viability, careful selection of structuring techniques and thermal management strategies is essential for maintaining stability and functionality over extended periods.

Industrial applications such as carbon capture and water harvesting require reticular materials to function under humid/ aqueous environments. However, exposure to moisture and acidic conditions often leads to hydrolysis and structural degradation in most MOFs.²²⁹ Post-synthetic modifications, such as fluorination or hydrophobic coatings, have been shown to enhance the water stability of MOFs while maintaining their adsorption properties.²³⁰ The structuring process may also influence hydrothermal stability; for example, blending MOFs into a polymer matrix may improve resistance to water-induced degradation compared to pristine MOF powders due to the protective polymer layer.⁹⁶ In contrast, COFs generally exhibit greater water stability due to their robust covalent linkages, yet their structured forms, such as membranes, can still be prone to swelling and pore blockage under prolonged water exposure. Advances in cross-linked COF membranes have addressed some of these issues, enhancing their thermal stability, mechanical stability and anti-swelling properties.²³¹ Furthermore, hybridization with hydrophobic fillers such as graphene has been explored to improve moisture tolerance in structured COFs.²³²

Despite the progress in improving the robustness of structured reticular materials, several unresolved challenges remain, and as such, future research should focus on: (i) developing advanced structuring techniques that preserve porosity while enhancing framework stability under extreme conditions; methods using standard unit operations will ensure low capex; (ii) investigating long-term performance through accelerated aging studies that simulate real-world industrial conditions; understanding the degradation pathways of structured materials will enable the design of more resilient frameworks;²³³ and (iii) establishing standardized testing protocols for evaluating the stability of structured reticular materials, which will facilitate the direct comparison of structuring techniques and accelerate their transition to commercial use.

5.2. Structured hybrid/composite materials

The integration of structured composite materials - where reticular frameworks are combined with polymers, metal oxides, or carbon-based supports - has been receiving attention as a means to enhance durability, scalability, and functional performance. For instance, MOF-polymer composites are being investigated due to their ability to retain the porosity and selectivity of MOFs while improving mechanical robustness and ease of processing.²³⁴ Polymer-supported MOF membranes and beads have demonstrated enhanced durability for gas separation, water purification, and catalysis.²³⁴ Studies have shown that polymer-grafted MOFs exhibit superior flexibility and adhesion properties, making them ideal candidates for coatings and sensor applications. Similarly, MOF-polymer aerogels have been developed for high-performance adsorption and catalysis due to their lightweight structure and large accessible surface area.134 The incorporation of metal and metal-oxide nanoparticles into MOFs has led to the realization of core-shell hybrid structures, enhancing catalytic and sensing properties. For instance, the encapsulation of Pd, Pt, and Au nanoparticles within MOFs has resulted in improved catalytic efficiency for hydrogenation and oxidation reactions.²³⁵ Meanwhile, SnO₂@MOF hybrids have exhibited significant performance in gas sensing due to the synergistic interactions between the porous architecture of MOFs and the semiconductor properties of SnO2.236 Additionally, magnetic MOF hybrids have been explored for their applications in environmental remediation, where their magnetic properties allow for easy separation and recovery after use.237

A key challenge in industrial applications of porous materials is their handling in large-scale reactors. MOF-alumina composites have demonstrated enhanced mechanical strength, making them suitable for fixed-bed adsorption systems.²³⁸ Similarly, carbon-based MOF films, such as UTSA-16/carbon hybrids, offer improved CO₂ adsorption capacity and faster adsorption kinetics – expanding their applicability in gas storage and separation.²³⁹ Developments in spray-dried MOF– polymer hybrids have enabled the formation of spherical granules with uniform porosity, facilitating their direct use in adsorption and catalysis.

A key issue in MOF composites is the optimization of interfacial interactions between MOFs and their composite phases to prevent phase separation and degradation. Fabrication techniques, such as ALD and *in situ* polymerization, are being explored to achieve better compatibility and stability. Looking ahead, the integration of machine learning and computational modeling in the design of structured composite materials will be important for optimizing processing conditions and predicting material stability. Furthermore, the adoption of sustainable and solvent-free processing techniques will help in aligning MOF-based composites with green manufacturing principles, promoting their broader adoption across industrial sectors.

5.3. Defects and their potential industrial implications

Structural defects may arise during shaping and processing and consequently may significantly influence their industrial performance. For example, mechanical compression, extrusion, solvent evaporation, additive manufacturing, and thermal processing introduce grain boundaries, fractures, voids, and partial amorphization - as discussed in previous sections - which can alter porosity, adsorption efficiency, and mechanical stability. While defects can compromise functionality, controlled defect engineering has, at times, been used as a tool to enhance performance.²⁴⁰ For instance, structural defects in UiO-66-NH₂ xerogels enhance CO₂ adsorption and separation kinetics, while controlled linker vacancies in catalytic MOFs boost efficiency.^{241,242} In conductive MOFs, defect pathways can facilitate electron mobility, while targeted defect introduction improves hydrothermal stability, preventing framework degradation in harsh environments.^{120,243} Scalable defect control, however, remains a challenge, requiring predictive computational models such as density functional theory (DFT) and machine learning for rational defect design. Long-term stability of defect-engineered materials must again be evaluated through accelerated aging studies and real-time process monitoring.

5.4. Cost reduction and sustainable structuring

For structured reticular materials to be widely adopted in industry, shaping techniques must be cost-effective, scalable, and compatible with existing manufacturing processes. Traditional methods like extrusion, pelletization, and granulation remain the most practical, offering high throughput and continuous processing with well-controlled mechanical properties. Compared to spray drying, which has high energy demands and material losses, extrusion is a more economical and scalable solution for producing mechanically stable structures. Incorporating biopolymer-based binders like alginate or cellulose

provides an eco-friendly alternative to synthetic additives, enhancing sustainability and mechanical performance while maintaining porosity as high as possible. Granulation and pelletization ensure high-scale production with sufficient mechanical strength for packed-bed applications. Sol-gel synthesis also offers advantages, particularly in producing monoliths with high mechanical resistance and controlled porosity. These monoliths can be molded into various shapes while maintaining structural integrity, making them useful for adsorption, catalysis, and separation applications. Their high density and mechanical stability allow direct use in gas storage and separation without additional shaping steps. However, optimizing sol-gel processes for large-scale production remains a challenge due to potential shrinkage and cracking. While 3D printing allows for complex geometries with tunable porosity, it currently remains impractical for industrial-scale production due to high capital expenditures, slow processing rates, and expensive feedstock. Unlike extrusion, which supports bulk material processing, 3D printing operates in batch mode, making it unsuitable for high-volume applications. Though beneficial for prototyping, the economic and technical barriers to scaling additive manufacturing outweigh its advantages for mainstream industrial use. Mechanochemical synthesis and shaping has been explored as a solvent-free alternative, but it struggles to match the structural integrity and porosity control achieved by wet chemistry using green solvents and is not a universal method for every material. While liquid-free processing reduces waste and improves sustainability, optimizing wetchemistry-based shaping with environmentally friendly solvents remains - probably - the more reliable and scalable approach. Industrial adoption also depends on addressing regulatory concerns, including compliance with environmental regulations, toxicity assessments, and stability requirements. Integrating lifecycle assessments, solvent recovery, and sustainable waste management into production workflows will ensure alignment with evolving regulations.

6. Outlook

Decades of research have positioned porous reticular materials at the forefront of adsorption-based energy applications, showing promising performance for handling challenges posed by new gas storage, separation, catalysis, and sensing-based applications. Our understanding of these materials - especially MOFs and COFs - have matured to the stage where we have the capability of establishing a high-degree of control over the structure, composition, functionality, and porosity of these materials for the application under consideration. Despite excellent performance metrics and promise, the use of reticular materials is - with some exceptions - largely limited to academic pursuits. While there are several components responsible for this translational gap, we have identified here the structuring and densification of these materials as a critical bottleneck. We have discussed progress in the structuring of porous reticular materials and have provided comprehensive

insights into tackling the challenges posed by the poor structuring of these materials. We have highlighted principles governing the hierarchical synthesis of these materials at the microscale, emphasizing the importance of establishing strong control over the quality of the material. In turn, we have presented the landscape of shapes and techniques available for structuring porous reticular materials at the macroscale, highlighting their unique advantages and disadvantages (Fig. 21).

Among the different shaping techniques, sol-gel synthesis stands out as a versatile method offering precise control over pore size and structure. In gas storage applications, the resulting high densities of sol-gel-shaped materials translate to remarkable volumetric capacities, crucial for confined spaces where maximizing gas storage within a limited volume is essential. Of course, this not only applies to gas storage but also to gas separation, where minimising the footprint is critical to reduce capital and operation costs. Having said that, there are certain challenges that exist enroute to industrial translation. If not done adequately, the high packing density of these materials can result in low diffusion coefficients for reactants and adsorbates, which may hinder performance in dynamic processes. Notably, in some cases, the adsorption kinetics can be higher in monoliths due to the better heat conductivity of densified bodies compared to powder.¹⁵⁸ Additionally, the synthesis of large-scale sol-gel materials may encounter issues pertaining to shrinkage and cracking - limiting scalability for mass production.

Dip coating is particularly suitable for creating thin films on complex shapes. It is well-suited for applications requiring effective sensors and membranes. In gas sensing, dip coating enables the deposition of sensitive sensing materials onto sensor substrates, facilitating the selective detection of target gases. However, controlling film thickness and porosity can pose challenges, potentially affecting sensor performance and reliability. Additionally, deposition techniques, such as PVD and CVD and their variants, offer precise control over film thickness, making them ideal for gas separation membranes. They allow for the deposition of ultra-thin films with customized properties, enabling efficient gas separation processes. However, implementing deposition techniques often requires specialized equipment and can be complex and costly.

Careful optimization of spray drying allows producing porous particles with a uniform size distribution. However, challenges arise in controlling pore structure and preventing agglomeration, which can impact material performance. Agglomeration of particles during the drying process can lead to uneven distribution of pores and reduce the overall surface area, limiting the effectiveness of the material in gas storage applications. On the other hand, extrusion is cost-effective and suitable for large-scale production of porous reticular materials, particularly for applications requiring catalyst supports and gas separation adsorbents. However, compared to other structuring methods, these techniques may result in reduced porosity and surface area as well as gas diffusivity due to compaction during mechanical compression. Despite these

Review Article



Fig. 21 A 'blueprint' summarising the key advantages and disadvantages, along with potential industrial applications of the main structuring techniques discussed in the review.

drawbacks, their simplicity and cost-effectiveness make them attractive options for industrial-scale production of materials.

3D printing stands out for its ability to fabricate complex geometries and tailor pore structures with precision, making it highly advantageous for various applications. In gas sensing, 3D printing allows for the creation of custom sensor designs with specific sensitivity and selectivity. By controlling the layout and composition of sensor components, it enables the development of sensors optimized for detecting target gases in specific environments. However, the adoption of 3D printing in gas sensing may be hindered by the need for specialized equipment and materials, as well as potential limitations in resolution compared to conventional bottom-up approaches as well as limitations in scalability. If cost is justified, the

flexibility and customization offered by 3D printing make it a promising technique for advancing gas sensing technology. Phase inversion and hydrogelation techniques provide versatile approaches for producing porous structures, encompassing both membranes and beads tailored for adsorbent and catalyst applications. These methods enable the controlled phase separation of polymer solutions or hydrogels, facilitating the creation of membranes with precisely controlled pore sizes and distributions. Widely utilized in membrane fabrication, these membranes exhibit tuneable characteristics ideal for efficient gas separation based on differences in molecular size and affinity. However, issues such as batch-to-batch variability, non-uniform bead sizes, and difficulties in achieving consistent pore structures pose obstacles to widespread industrial adoption. Despite these challenges, ongoing research efforts aim to address these limitations and enhance the scalability and reliability of phase inversion and hydrogelation techniques for industrial-scale production. These efforts include exploring novel methods for process control, advanced characterization techniques, and the development of more robust materials to overcome the current limitations and accelerate the industrial adoption of these promising techniques.

A key avenue for overcoming these challenges lies in computational modeling and simulations, which provide predictive insights into structuring methodologies. Traditional experimental approaches to structuring reticular materials often rely on iterative trial-and-error processes, which are timeconsuming and inefficient. The integration of density functional theory (DFT), molecular dynamics (MD), and finite element modeling (FEM) has allowed researchers to simulate the effects of mechanical compression, extrusion, and thermal processing on porosity, mechanical stability, and adsorption behavior. Computational fluid dynamics (CFD) simulations further enable the study of mass transfer and diffusion properties within structured monoliths, membranes, and aerogels-critical for optimizing gas separation and catalytic applications. Beyond predicting processing outcomes, computational approaches also aid in defect engineering in shaped materials. While structuring techniques can introduce defects such as missing linkers, grain boundaries, or pore collapse, simulations provide a means to predict, control, and even leverage these defects for enhanced functionality. Furthermore, machine learning algorithms trained on large experimental datasets can optimize shaping parameters such as pressure, binder content, and solvent evaporation rates, enabling faster and more efficient material development.

Despite the progress made in structuring porous reticular materials, several key challenges must be addressed before their large-scale industrial adoption becomes feasible. One of the most pressing concerns is the ability to maintain structural integrity and porosity throughout shaping and densification. Many conventional shaping techniques compromise essential properties such as surface area, pore connectivity, and mechanical stability. Although recent advancements in additive manufacturing, templated synthesis, and sol–gel methods have demonstrated promise, further optimization is required to ensure these techniques preserve functional properties across both micro- and macroscales. Future research should focus on developing shaping strategies that strike a balance between densification and porosity retention, minimizing pore collapse while maintaining sufficient mechanical strength and stability for industrial applications.

Another major challenge lies in the development of scalable and cost-effective structuring methods. Advanced techniques such as 3D printing and chemical vapor deposition offer precise control over material structuring, but their reliance on expensive precursors and specialized equipment limits their feasibility for large-scale production. Furthermore, understanding the impact of processing parameters – such as pressure, temperature, and precursor concentration – on the final properties of shaped materials will be crucial in establishing robust and reproducible structuring methodologies.

Long-term stability is another critical concern, as structured MOFs and COFs must withstand real-world operational conditions, including fluctuations in humidity, temperature, and mechanical stress. While much research has been dedicated to optimizing initial material performance, less is known about how these materials degrade over time in industrial harsh environments. Degradation mechanisms such as framework collapse, chemical instability, and fouling must be systematically investigated. Future efforts should prioritize accelerated aging studies, *in situ* characterization techniques, and the development of protective coatings to enhance material durability, ensuring their reliability for practical applications.

In addition to experimental advancements, improved theoretical and computational models are essential for guiding the structuring of porous materials. Existing models often focus on idealized structures, whereas real-world applications involve complex morphologies and heterogeneous environments. The integration of machine learning and computational simulations could provide predictive insights into key parameters such as pore connectivity, mechanical resilience, and diffusion efficiency, allowing for the rational design of structured MOFs and COFs. By coupling experimental research with computational tools, researchers can accelerate the development of highly functional, structured reticular materials tailored for specific industrial needs.

Finally, sustainability and recyclability must be at the forefront of future research in structured reticular materials. As industries transition towards circular economy principles, it is critical to develop environmentally friendly synthesis routes that minimize solvent use, energy consumption, and waste generation. Additionally, research into the regeneration, reuse, and recyclability of shaped MOFs and COFs will be crucial for ensuring long-term sustainability. Addressing these challenges will facilitate the transition of structured porous materials from niche academic research to widespread industrial applications, ultimately playing a transformative role in energy storage, environmental remediation, and next-generation catalysis.

Looking at a broader perspective, there is a clear need for developing a deeper fundamental appreciation of how properties at the microscale influence the behaviour at the macroscale. This appreciation would serve to guide a rational, judicious selection of structuring techniques for the effective implementation of clearly promising materials at industrial scales. Here, a judicious combination of the techniques that we have discussed in the present context may also hold promise. As the development of these materials advances, it is a sincere hope that their structuring is not forgotten in the process.

Data availability

No primary research results and no new data were generated or analysed in the context of this review article.

Conflicts of interest

D. F.-J. has a financial interest in the start-up company Immaterial, which is seeking to commercialize metal-organic frameworks.

Acknowledgements

The authors thank funding from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme (NanoMOFdeli), ERC-2016-COG 726380, and the EPSRC (EP/S009000/1). M. A. acknowledges funding by UK Research and Innovation (UKRI) under the UK government's Horizon Europe funding guarantee (EP/Y023447/1). A. P. acknowledges the financial support of the "Margarita Salas" postdoctoral program funded by the Ministerio de Universidades, and the European Union – Next Generation (CA1/RSUE/2021-0071) from the Spanish Ministerio de Universidades and Universidad Autónoma de Madrid. D. M. acknowledges NanoDTC Cambridge – EPSRC EP/S022953/1.

References

- 1 M. E. Davis, Nature, 2002, 417, 813-821.
- 2 S. Kitagawa, Angew. Chem., Int. Ed., 2015, 54, 10686-10687.
- 3 W. Xu, B. Tu, Q. Liu, Y. Shu, C.-C. Liang, C. S. Diercks, O. M. Yaghi, Y.-B. Zhang, H. Deng and Q. Li, *Nat. Rev. Mater.*, 2020, 5, 764–779.
- 4 H. Jiang, D. Alezi and M. Eddaoudi, *Nat. Rev. Mater.*, 2021, 6, 466–487.
- 5 H. Jiang, S. Benzaria, N. Alsadun, J. Jia, J. Czaban-Jóźwiak,
 V. Guillerm, A. Shkurenko, Z. Thiam, M. Bonneau,
 V. K. Maka, Z. Chen, Z. O. Ameur, M. O'Keeffe and
 M. Eddaoudi, *Science*, 2024, 386, 659–666.
- 6 O. M. Yaghi, ACS Cent. Sci., 2019, 5, 1295-1300.
- 7 H. Furukawa, K. E. Cordova, M. O'Keeffe and O. M. Yaghi, *Science*, 2013, **341**, 1230444.
- 8 S. Kitagawa, R. Kitaura and S. Noro, *Angew. Chem., Int. Ed.*, 2004, **43**, 2334–2375.
- 9 X. Feng, X. Ding and D. Jiang, *Chem. Soc. Rev.*, 2012, 41, 6010–6022.

- 10 S. Lee, H. Jeong, D. Nam, M. S. Lah and W. Choe, *Chem. Soc. Rev.*, 2021, **50**, 528–555.
- 11 P. J. Waller, F. Gándara and O. M. Yaghi, *Acc. Chem. Res.*, 2015, **48**, 3053–3063.
- 12 J. D. Evans, C. J. Sumby and C. J. Doonan, *Chem. Lett.*, 2015, 44, 582–588.
- 13 R.-B. Lin and B. Chen, Chem, 2022, 8, 2114-2135.
- 14 P. Z. Moghadam, A. Li, S. B. Wiggin, A. Tao, A. G. P. Maloney, P. A. Wood, S. C. Ward and D. Fairen-Jimenez, *Chem. Mater.*, 2017, 29, 2618–2625.
- 15 A. Khobotov-Bakishev, L. Hernández-López, C. von Baeckmann, J. Albalad, A. Carné-Sánchez and D. Maspoch, *Adv. Sci.*, 2022, **9**, 2104753.
- 16 B. M. Connolly, D. G. Madden, A. E. H. Wheatley and D. Fairen-Jimenez, J. Am. Chem. Soc., 2020, 142, 8541–8549.
- 17 L. J. Murray, M. Dincă and J. R. Long, *Chem. Soc. Rev.*, 2009, 38, 1294.
- 18 T. Tian, Z. Zeng, D. Vulpe, M. E. Casco, G. Divitini, P. A. Midgley, J. Silvestre-Albero, J.-C. Tan, P. Z. Moghadam and D. Fairen-Jimenez, *Nat. Mater.*, 2018, 17, 174–179.
- 19 Y. Lin, W. Li, Y. Wen, G. Wang, X. Ye and G. Xu, Angew. Chem., Int. Ed., 2021, 60, 25758–25761.
- 20 J. Y. S. Lin, Science, 2016, 353, 121-122.
- 21 M. S. Denny, J. C. Moreton, L. Benz and S. M. Cohen, *Nat. Rev. Mater.*, 2016, **1**, 16078.
- 22 R. V. Jagadeesh, K. Murugesan, A. S. Alshammari, H. Neumann, M.-M. Pohl, J. Radnik and M. Beller, *Science*, 2017, **358**, 326–332.
- 23 A. Bétard and R. A. Fischer, *Chem. Rev.*, 2012, **112**, 1055–1083.
- 24 H. Li, K. Wang, Y. Sun, C. T. Lollar, J. Li and H.-C. Zhou, *Mater. Today*, 2018, **21**, 108–121.
- 25 Q. Qian, P. A. Asinger, M. J. Lee, G. Han, K. Mizrahi Rodriguez, S. Lin, F. M. Benedetti, A. X. Wu, W. S. Chi and Z. P. Smith, *Chem. Rev.*, 2020, **120**, 8161–8266.
- 26 D. Yang and B. C. Gates, ACS Catal., 2019, 9, 1779-1798.
- 27 A. M. Wright, M. T. Kapelewski, S. Marx, O. K. Farha and W. Morris, *Nat. Mater.*, 2025, 24, 178–197.
- 28 R. Freund, O. Zaremba, G. Arnauts, R. Ameloot, G. Skorupskii, M. Dincă, A. Bavykina, J. Gascon, A. Ejsmont, J. Goscianska, M. Kalmutzki, U. Lächelt, E. Ploetz, C. S. Diercks and S. Wuttke, *Angew. Chem., Int. Ed.*, 2021, 60, 23975–24001.
- 29 E. Maine and P. Seegopaul, Nat. Mater., 2016, 15, 487-491.
- 30 Svante: Carbon Capture Technology for a Sustainable Future, https://www.svanteinc.com/carbon-capture-technology/, (accessed 13 March 2025).
- 31 Our Technology Immaterial, https://immaterial.com/tech nology/, (accessed 13 March 2025).
- 32 J. Bin Lin, T. T. T. Nguyen, R. Vaidhyanathan, J. Burner, J. M. Taylor, H. Durekova, F. Akhtar, R. K. Mah, O. Ghaffari-Nik, S. Marx, N. Fylstra, S. S. Iremonger, K. W. Dawson, P. Sarkar, P. Hovington, A. Rajendran, T. K. Woo and G. K. H. Shimizu, *Science*, 2021, 374, 1464–1469.
- 33 A. J. Rieth, A. M. Wright and M. Dincă, Nat. Rev. Mater., 2019, 4, 708–725.

- 34 T. D. Bennett and S. Horike, *Nat. Rev. Mater.*, 2018, 3, 431-440.
- 35 T. D. Bennett, P. Simoncic, S. A. Moggach, F. Gozzo, P. Macchi, D. A. Keen, J.-C. Tan and A. K. Cheetham, *Chem. Commun.*, 2011, 47, 7983.
- 36 P. Z. Moghadam, S. M. J. Rogge, A. Li, C.-M. Chow, J. Wieme, N. Moharrami, M. Aragones-Anglada, G. Conduit, D. A. Gomez-Gualdron, V. Van Speybroeck and D. Fairen-Jimenez, *Matter*, 2019, **1**, 219–234.
- 37 L. Shi, Y. Zhong, H. Cao, H. Wang, Z. Xiong, K. Wang,
 H. Shen and Z. Chen, *Nat. Synth.*, 2024, 3, 1560–1566.
- 38 S. Furukawa, J. Reboul, S. Diring, K. Sumida and S. Kitagawa, *Chem. Soc. Rev.*, 2014, **43**, 5700–5734.
- 39 N. Stock and S. Biswas, Chem. Rev., 2012, 112, 933-969.
- 40 N. T. K. Thanh, N. Maclean and S. Mahiddine, *Chem. Rev.*, 2014, **114**, 7610–7630.
- 41 V. V. Karve, A. Mabillard, J. Espin Marti, M. Asgari, W. L. Queen and M. Soutrenon, *Mater. Res. Express*, 2024, 11, 055511.
- 42 J. J. Purewal, D. Liu, J. Yang, A. Sudik, D. J. Siegel, S. Maurer and U. Müller, *Int. J. Hydrogen Energy*, 2012, 37, 2723–2727.
- 43 C. R. Marshall, S. A. Staudhammer and C. K. Brozek, *Chem. Sci.*, 2019, **10**, 9396–9408.
- 44 V. K. Lamer and R. H. Dinegar, J. Am. Chem. Soc., 1950, 72, 4847–4854.
- 45 S. Hermes, T. Witte, T. Hikov, D. Zacher, S. Bahnmüller, G. Langstein, K. Huber and R. A. Fischer, *J. Am. Chem. Soc.*, 2007, **129**, 5324–5325.
- 46 P. Schäfer, F. Kapteijn, M. A. Van Der Veen and K. F. Domke, *Cryst. Growth Des.*, 2017, **17**, 5603–5607.
- 47 J. Park, Q. Jiang, D. Feng, L. Mao and H.-C. Zhou, J. Am. Chem. Soc., 2016, 138, 3518–3525.
- 48 W. Ma, Q. Zheng, Y. He, G. Li, W. Guo, Z. Lin and L. Zhang, J. Am. Chem. Soc., 2019, 141, 18271–18277.
- 49 R. S. Forgan, Chem. Sci., 2020, 11, 4546-4562.
- 50 J. Łuczak, M. Kroczewska, M. Baluk, J. Sowik, P. Mazierski and A. Zaleska-Medynska, *Adv. Colloid Interface Sci.*, 2023, **314**, 102864.
- 51 H. Guo, Y. Zhu, S. Wang, S. Su, L. Zhou and H. Zhang, *Chem. Mater.*, 2012, **24**, 444–450.
- 52 T. Tsuruoka, S. Furukawa, Y. Takashima, K. Yoshida,
 S. Isoda and S. Kitagawa, *Angew. Chem., Int. Ed.*, 2009,
 48, 4739–4743.
- 53 H. Guo, Y. Zhu, S. Qiu, A. J. Lercher and H. Zhang, Adv. Mater., 2010, 22, 4190–4192.
- 54 F. Wang, H. Guo, Y. Chai, Y. Li and C. Liu, *Microporous Mesoporous Mater.*, 2013, **173**, 181–188.
- 55 D. Lyu, W. Xu, J. E. L. Payong, T. Zhang and Y. Wang, *Nat. Commun.*, 2022, **13**, 3980.
- 56 J. Xiao, J. Chen and H. Qiu, *Green Chem.*, 2022, 24, 2193–2202.
- 57 E. Ploetz, H. Engelke, U. Lächelt and S. Wuttke, *Adv. Funct. Mater.*, 2020, **30**, 1909062.
- 58 M. L. Hu, M. Y. Masoomi and A. Morsali, *Coord. Chem. Rev.*, 2019, 387, 415–435.

- 59 J. Tang and Y. Yamauchi, Nat. Chem., 2016, 8, 638-639.
- 60 H. Kim and M. S. Lah, Dalton Trans., 2017, 46, 6146-6158.
- 61 Y. Liu, J. Goebl and Y. Yin, *Chem. Soc. Rev.*, 2013, 42, 2610–2653.
- 62 H. Huang, J. R. Li, K. Wang, T. Han, M. Tong, L. Li, Y. Xie, Q. Yang, D. Liu and C. Zhong, *Nat. Commun.*, 2015, 6, 8847.
- 63 J. Wang, J. Tang, B. Ding, Z. Chang, X. Hao, T. Takei, N. Kobayashi, Y. Bando, X. Zhang and Y. Yamauchi, *Small*, 2018, 14, 1704461.
- 64 J. Yang, K. Li and J. Gu, ACS Mater. Lett., 2022, 4, 385-391.
- 65 S. Li, W. Zhang, Y. Zhu, Q. Zhao and F. Huo, *Cryst. Growth Des.*, 2015, 15, 1017–1021.
- 66 N. Moitra, S. Fukumoto, J. Reboul, K. Sumida, Y. Zhu, K. Nakanishi, S. Furukawa, S. Kitagawa and K. Kanamori, *Chem. Commun.*, 2015, **51**, 3511–3514.
- 67 K. Li, J. Yang and J. Gu, Acc. Chem. Res., 2022, 55, 2235-2247.
- 68 S. Begum, Z. Hassan, S. Bräse, C. Wöll and M. Tsotsalas, *Acc. Chem. Res.*, 2019, **52**, 1598–1610.
- 69 E. Doustkhah, R. Hassandoost, A. Khataee, R. Luque and M. H. N. Assadi, *Chem. Soc. Rev.*, 2021, 50, 2927–2953.
- 70 F. Zhang, Y. Wei, X. Wu, H. Jiang, W. Wang and H. Li, J. Am. Chem. Soc., 2014, 136, 13963–13966.
- 71 K. Shen, L. Zhang, X. Chen, L. Liu, D. Zhang, Y. Han, J. Chen, J. Long, R. Luque, Y. Li and B. Chen, *Science*, 2018, 359, 206–210.
- 72 H. J. Lee, W. Cho and M. Oh, *Chem. Commun.*, 2012, 48, 221–223.
- 73 Z. Karimi and A. Morsali, J. Mater. Chem. A, 2013, 1, 3047–3054.
- 74 G. Cai, W. Zhang, L. Jiao, S. H. Yu and H. L. Jiang, *Chem*, 2017, 2, 791–802.
- 75 F. L. Li, H. X. Li and J. P. Lang, *CrystEngComm*, 2016, 18, 1760–1767.
- 76 X. Wu, J. Lin, Y. He, R. Wang, Y. Zhou, H. Qi, Y. Chen, P. Zuo, X. Liu and Q. Zhuang, *Carbon*, 2024, 227, 119289.
- 77 D. E. Kravchenko, A. J. Cruz, S. Rodríguez-Hermida, N. Wauteraerts, T. Hauffman and R. Ameloot, *Chem. Mater.*, 2020, 32, 10469–10475.
- 78 T. Xu, K. Sun, D. Gao, C. Li, X. Hu and G. Chen, *Chem. Commun.*, 2019, 55, 7651–7654.
- 79 K. Li, J. Yang, R. Huang, S. Lin and J. Gu, Angew. Chem., Int. Ed., 2020, 59, 14124–14128.
- 80 L. Zhao, J. Yang, M. Gong, K. Li and J. Gu, J. Am. Chem. Soc., 2021, 143, 15145–15151.
- 81 K. Li, Y. Zhao, J. Yang and J. Gu, *Nat. Commun.*, 2022, 13, 1879.
- 82 S. Furukawa, K. Hirai, K. Nakagawa, Y. Takashima, R. Matsuda, T. Tsuruoka, M. Kondo, R. Haruki, D. Tanaka, H. Sakamoto, S. Shimomura, O. Sakata and S. Kitagawa, *Angew. Chem., Int. Ed.*, 2009, **48**, 1766–1770.
- 83 Y. Y. Liu, X. C. Li, S. Wang, T. Cheng, H. Yang, C. Liu,
 Y. Gong, W. Y. Lai and W. Huang, *Nat. Commun.*, 2020,
 11, 5561.
- 84 S. Kandambeth, V. Venkatesh, D. B. Shinde, S. Kumari,
 A. Halder, S. Verma and R. Banerjee, *Nat. Commun.*, 2015,
 6, 6786.

- 85 E. Jin, K. Geng, K. H. Lee, W. Jiang, J. Li, Q. Jiang, S. Irle and D. Jiang, *Angew. Chem.*, *Int. Ed.*, 2020, **59**, 12162–12169.
- 86 C. Franco, D. Rodríguez-San-Miguel, A. Sorrenti, S. Sevim, R. Pons, A. E. Platero-Prats, M. Pavlovic, I. Szilágyi, M. L. Ruiz Gonzalez, J. M. González-Calbet, D. Bochicchio, L. Pesce, G. M. Pavan, I. Imaz, M. Cano-Sarabia, D. Maspoch, S. Pané, A. J. De Mello, F. Zamora and J. Puigmartí-Luis, *J. Am. Chem. Soc.*, 2020, **142**, 3540–3547.
- 87 T. Hasell and A. I. Cooper, Nat. Rev. Mater., 2016, 1, 1-14.
- 88 J. Fonseca, L. Meng, I. Imaz and D. Maspoch, *Chem. Soc. Rev.*, 2023, **52**, 2528–2543.
- 89 J. Wang, Y. Liu, G. Bleyer, E. S. A. Goerlitzer, S. Englisch, T. Przybilla, C. F. Mbah, M. Engel, E. Spiecker, I. Imaz, D. Maspoch and N. Vogel, *Angew. Chem.*, 2022, **134**, e202117455.
- 90 N. Yanai, M. Sindoro, J. Yan and S. Granick, J. Am. Chem. Soc., 2013, 135, 34–37.
- 91 N. Yanai and S. Granick, *Angew. Chem., Int. Ed.*, 2012, **51**, 5638–5641.
- 92 G. Lu, C. Cui, W. Zhang, Y. Liu and F. Huo, *Chem. Asian J.*, 2013, 8, 69–72.
- 93 C. Avci, I. Imaz, A. Carné-Sánchez, J. A. Pariente, N. Tasios,
 J. Pérez-Carvajal, M. I. Alonso, A. Blanco, M. Dijkstra,
 C. López and D. Maspoch, *Nat. Chem.*, 2018, 10, 78–84.
- 94 C. Avci, Y. Liu, J. A. Pariente, A. Blanco, C. Lopez, I. Imaz and D. Maspoch, *Small*, 2019, **15**, 1902520.
- 95 B. Gole, V. Stepanenko, S. Rager, M. Grüne, D. D. Medina, T. Bein, F. Würthner and F. Beuerle, *Angew. Chem., Int. Ed.*, 2018, 57, 846–850.
- 96 Q. Ma, T. Zhang and B. Wang, Matter, 2022, 5, 1070-1091.
- 97 L. Shi, Z. Xiong, H. Wang, H. Cao and Z. Chen, *Chem*, 2024, 10, 2464–2472.
- 98 E. Adatoz, A. K. Avci and S. Keskin, Sep. Purif. Technol., 2015, 152, 207–237.
- 99 J. Hou, A. F. Sapnik and T. D. Bennett, *Chem. Sci.*, 2020, **11**, 310–323.
- 100 X. Huang, X. Chen, A. Li, D. Atinafu, H. Gao, W. Dong and G. Wang, *Chem. Eng. J.*, 2019, 356, 641–661.
- 101 M. Ding, X. Cai and H.-L. Jiang, Chem. Sci., 2019, 10, 10209-10230.
- 102 A. G. Slater and A. I. Cooper, *Science*, 2015, 348, aaa8075.
- 103 P. Liu and G. F. Chen, *Porous Materials: Processing and Applications*, Elsevier Science, 2014.
- 104 S. E. M. Elhenawy, M. Khraisheh, F. AlMomani and G. Walker, *Catalysts*, 2020, **10**, 1293.
- Y. N. Wu, Y. Fang, J. Fu, L. He, D. M. Kabtamu, L. Matovic,
 F. Li and J. Li, *J. Environ. Chem. Eng.*, 2022, 10, 108556.
- F. Figueira, R. F. Mendes, E. M. Domingues, P. Barbosa,
 F. Figueiredo, F. A. A. Paz and J. Rocha, *Appl. Sci.*, 2020,
 10, 798.
- 107 B. M. Connolly, M. Aragones-Anglada, J. Gandara-Loe, N. A. Danaf, D. C. Lamb, J. P. Mehta, D. Vulpe, S. Wuttke, J. Silvestre-Albero, P. Z. Moghadam, A. E. H. Wheatley and D. Fairen-Jimenez, *Nat. Commun.*, 2019, **10**, 1–11.
- 108 S. Fajal, W. Mandal, A. Torris, D. Majumder, S. Let, A. Sen, F. Kanheerampockil, M. M. Shirolkar and S. K. Ghosh, *Nat. Commun.*, 2024, **15**, 1278.

- 109 X. Luo, M. Zhang, Y. Hu, Y. Xu, H. Zhou, Z. Xu, Y. Hao, S. Chen, S. Chen, Y. Luo, Y. Lin and J. Zhao, *Science*, 2024, 385, 647–651.
- 110 W. Y. Hong, S. P. Perera and A. D. Burrows, *Microporous Mesoporous Mater.*, 2020, **308**, 110525.
- 111 W.-L. Xue, P. Kolodzeiski, H. Aucharova, S. Vasa,
 A. Koutsianos, R. Pallach, J. Song, L. Frentzel-Beyme,
 R. Linser and S. Henke, *Nat. Commun.*, 2024, 15, 4420.
- 112 A. H. Valekar, K.-H. Cho, U.-H. Lee, J. S. Lee, J. W. Yoon,
 Y. K. Hwang, S. G. Lee, S. J. Cho and J.-S. Chang, *RSC Adv.*, 2017, 7, 55767–55777.
- 113 C. Perez-Rosales and J. J. Martinez, *Soc. Pet. Eng. J.*, 1971, **11**, 363–366.
- 114 A. Terzis, A. Ramachandran, K. Wang, M. Asheghi, K. E. Goodson and J. G. Santiago, *Cell Rep. Phys. Sci.*, 2020, **1**, 100057.
- 115 A. Santomaso, P. Lazzaro and P. Canu, *Chem. Eng. Sci.*, 2003, **58**, 2857–2874.
- 116 C. Yang, Y. Zhu, J. Wang, W. Sun, L. Yang, H. Lin and S. Lv, *Chem. Eng. J.*, 2021, **426**, 131724.
- 117 Z. Wang, L. Liu, Z. Li, N. Goyal, T. Du, J. He and G. K. Li, Energy Fuels, 2022, **36**, 2927–2944.
- 118 S. Afandizadeh and E. A. Foumeny, *Appl. Therm. Eng.*, 2001, 21, 669–682.
- B. Yeskendir, J.-P. Dacquin, Y. Lorgouilloux, C. Courtois, S. Royer and J. Dhainaut, *Mater. Adv.*, 2021, 2, 7139–7186.
- 120 J. Ren, N. M. Musyoka, H. W. Langmi, A. Swartbooi,
 B. C. North and M. Mathe, *Int. J. Hydrogen Energy*, 2015,
 40, 4617–4622.
- 121 J. Alemán, A. V. Chadwick, J. He, M. Hess, K. Horie, R. G. Jones, P. Kratochvíl, I. Meisel, I. Mita, G. Moad, S. Penczek and R. F. T. Stepto, *Pure Appl. Chem.*, 2007, 79, 1801–1829.
- 122 A. Ahmed, M. Förster, R. Clowes, P. Myers and H. Zhang, *Chem. Commun.*, 2014, **50**, 14314–14316.
- 123 J. L. Williams, Catal. Today, 2001, 69, 3-9.
- 124 S. A. El-Safty, J. Porous Mater., 2011, 18, 259–287.
- 125 L. A. Darunte, Y. Terada, C. R. Murdock, K. S. Walton, D. S. Sholl and C. W. Jones, ACS Appl. Mater. Interfaces, 2017, 9, 17042–17050.
- 126 S. Lawson, A. A. Rownaghi and F. Rezaei, *Energy Technol.*, 2018, **6**, 694–701.
- 127 P. Albacete, M. Asgari, Y. Yang, A. N. Al-Shanks and D. Fairen-Jimenez, *Adv. Funct. Mater.*, 2024, **34**, 2305979.
- 128 L. Huang, J. Yang, Y. Zhao, H. Miyata, M. Han, Q. Shuai and Y. Yamauchi, *Chem. Mater.*, 2023, **35**, 2661–2682.
- 129 D. Zhu, Y. Zhu, Q. Yan, M. Barnes, F. Liu, P. Yu, C. P. Tseng, N. Tjahjono, P. C. Huang, M. M. Rahman, E. Egap, P. M. Ajayan and R. Verduzco, *Chem. Mater.*, 2021, 33, 4216–4224.
- 130 E. Candish, H.-J. Wirth, A. A. Gooley, R. A. Shellie and E. F. Hilder, *J. Chromatogr. A*, 2015, **1410**, 9–18.
- 131 T. Tian, J. Velazquez-Garcia, T. D. Bennett and D. Fairen-Jimenez, J. Mater. Chem. A, 2015, 3, 2999–3005.
- 132 D. Bokov, A. Turki Jalil, S. Chupradit, W. Suksatan,
 M. Javed Ansari, I. H. Shewael, G. H. Valiev and
 E. Kianfar, *Adv. Mater. Sci. Eng.*, 2021, 1, 5102014.

- 133 N. Brun, S. Ungureanu, H. Deleuze and R. Backov, *Chem. Soc. Rev.*, 2011, **40**, 771–788.
- 134 T. Guo, H. Mashhadimoslem, L. Choopani, M. M. Salehi, A. Maleki, A. Elkamel, A. Yu, Q. Zhang, J. Song, Y. Jin and O. J. Rojas, *Small*, 2024, **20**, 2402942.
- 135 Y. Chen, X. Huang, S. Zhang, S. Li, S. Cao, X. Pei, J. Zhou,
 X. Feng and B. Wang, *J. Am. Chem. Soc.*, 2016, 138, 10810–10813.
- 136 O. Shekhah, J. Liu, R. A. Fischer and C. Wöll, *Chem. Soc. Rev.*, 2011, 40, 1081.
- 137 X. Chen, J. Dong, K. Chi, L. Wang, F. Xiao, S. Wang, Y. Zhao and Y. Liu, *Adv. Funct. Mater.*, 2021, **31**, 2102855.
- 138 J.-H. Yim, M. R. Baklanov, D. W. Gidley, H. Peng, H.-D. Jeong and L. Sun Pu, *J. Phys. Chem. B*, 2004, **108**, 8953–8959.
- 139 I. Tokarev and S. Minko, Adv. Mater., 2010, 22, 3446-3462.
- 140 E. A. Jackson and M. A. Hillmyer, ACS Nano, 2010, 4, 3548–3553.
- 141 J.-S. Jang, L. R. Winter, C. Kim, J. D. Fortner and M. Elimelech, *Trends Chem.*, 2021, 3, 547–560.
- 142 N. Ma and S. Horike, Chem. Rev., 2022, 122, 4163-4203.
- 143 R. Lin, M. Chai, Y. Zhou, V. Chen, T. D. Bennett and J. Hou, *Chem. Soc. Rev.*, 2023, **52**, 4149–4172.
- 144 L. N. McHugh and T. D. Bennett, J. Mater. Chem. A, 2022, 10, 19552–19559.
- 145 C. Gao, Z. Jiang, S. Qi, P. Wang, L. R. Jensen, M. Johansen, C. K. Christensen, Y. Zhang, D. B. Ravnsbæk and Y. Yue, *Adv. Mater.*, 2022, 34, 2110048.
- 146 K. Tanaka, Y. Tago, M. Kondo, Y. Watanabe, K. Nishio, T. Hitosugi and M. Moriya, *Nano Lett.*, 2020, 20, 8200–8204.
- 147 Y. Wang, H. Jin, Q. Ma, K. Mo, H. Mao, A. Feldhoff, X. Cao,Y. Li, F. Pan and Z. Jiang, *Angew. Chem.*, 2020, 132, 4395–4399.
- 148 A. C. Pierre, *Introduction to Sol-Gel Processing*, Springer International Publishing, Cham, 2020, 421–455.
- 149 S. M. F. Vilela, P. Salcedo-Abraira, L. Micheron, E. L. Solla, P. G. Yot and P. Horcajada, *Chem. Commun.*, 2018, 54, 13088–13091.
- 150 S. Wang, C. M. McGuirk, A. d'Aquino, J. A. Mason and C. A. Mirkin, *Adv. Mater.*, 2018, **30**, 1800202.
- 151 W. Xia, J. Zhu, W. Guo, L. An, D. Xia and R. Zou, *J. Mater. Chem. A*, 2014, **2**, 11606–11613.
- 152 T. Chalati, P. Horcajada, R. Gref, P. Couvreur and C. Serre, *J. Mater. Chem.*, 2011, **21**, 2220–2227.
- 153 A. de la Peña Ruigómez, D. Rodríguez-San-Miguel, K. C. Stylianou, M. Cavallini, D. Gentili, F. Liscio, S. Milita, O. M. Roscioni, M. L. Ruiz-González, C. Carbonell, D. Maspoch, R. Mas-Ballesté, J. L. Segura and F. Zamora, *Chem. Eur. J.*, 2015, **21**, 10666–10670.
- 154 M. E. Carrington, N. Rampal, D. G. Madden, D. O'Nolan, N. P. M. Casati, G. Divitini, J. Martín-Illán, M. Tricarico, R. Cepitis, C. Çamur, T. Curtin, J. Silvestre-Albero, J. C. Tan, F. Zamora, S. Taraskin, K. W. Chapman and D. Fairen-Jimenez, *Chem*, 2022, 8, 2961–2977.
- 155 E. Tsalaporta and J. M. D. MacElroy, *Heliyon*, 2020, **6**, e04883.

- 156 D. G. Madden, D. O'Nolan, N. Rampal, R. Babu, C. Çamur, A. N. Al Shakhs, S.-Y. Zhang, G. A. Rance, J. Perez, N. Pietro Maria Casati, C. Cuadrado-Collados, D. O'Sullivan, N. P. Rice, T. Gennett, P. Parilla, S. Shulda, K. E. Hurst, V. Stavila, M. D. Allendorf, J. Silvestre-Albero, A. C. Forse, N. R. Champness, K. W. Chapman and D. Fairen-Jimenez, *J. Am. Chem. Soc.*, 2022, **144**, 13729–13739.
- 157 Y. Peng, V. Krungleviciute, I. Eryazici, J. T. Hupp, O. K. Farha and T. Yildirim, *J. Am. Chem. Soc.*, 2013, 135, 11887–11894.
- 158 C. Çamur, R. Babu, J. A. Suárez del Pino, N. Rampal, J. Pérez-Carvajal, P. Hügenell, S. Ernst, J. Silvestre-Albero, I. Imaz, D. G. Madden, D. Maspoch and D. Fairen-Jimenez, *Adv. Mater.*, 2023, 35, 2209104.
- 159 S. Fan, Z. Chen, Z. Yang, J. Feng, L. Yu, Z. Qiu, W. Liu, B. Li and S. Zhang, *AIChE J.*, 2022, **68**, e178172.
- 160 D. G. Madden, R. Babu, C. Çamur, N. Rampal, J. Silvestre-Albero, T. Curtin and D. Fairen-Jimenez, *Faraday Discuss.*, 2021, 231, 51–65.
- 161 J. Martín-Illán, D. Rodríguez-San-Miguel, O. Castillo,
 G. Beobide, J. Perez-Carvajal, I. Imaz, D. Maspoch and
 F. Zamora, *Angew. Chem., Int. Ed.*, 2021, 60, 13969–13977.
- 162 D. Fairen-Jimenez, S. A. Moggach, M. T. Wharmby, P. A. Wright, S. Parsons and T. Duren, *J. Am. Chem. Soc.*, 2011, **133**, 8900–8902.
- 163 J. P. Mehta, T. Tian, Z. Zeng, G. Divitini, B. M. Connolly, P. A. Midgley, J. Tan, D. Fairen-Jimenez and A. E. H. Wheatley, *Adv. Funct. Mater.*, 2018, 28, 1705588.
- 164 J.-W. Ye, X. Zhou, Y. Wang, R.-K. Huang, H.-L. Zhou, X.-N. Cheng, Y. Ma and J.-P. Zhang, *Sci. China Mater.*, 2018, **61**, 424–428.
- 165 T. Tian, J. Xu, A. Abdolazizi, C. Ji, J. Hou, D. J. Riley, C. Yan, M. P. Ryan, F. Xie and C. Petit, *Mater. Today NANO*, 2023, 21, 100293.
- 166 X. Tang and X. Yan, J. Sol-Gel Sci. Technol., 2017, 81, 379-404.
- 167 K. Mori, T. Fujita, H. Hata, H.-J. Kim, T. Nakano and H. Yamashita, ACS Appl. Mater. Interfaces, 2023, 15, 51079–51088.
- 168 E. Gkaniatsou, C. Chen, F. S. Cui, X. Zhu, P. Sapin, F. Nouar, C. Boissière, C. N. Markides, J. Hensen and C. Serre, *Cell Rep. Phys. Sci.*, 2022, 3, 100730.
- 169 L. Sarango, L. Paseta, M. Navarro, B. Zornoza and J. Coronas, J. Ind. Eng. Chem., 2018, **59**, 8–16.
- 170 G. Li, K. Zhang and T. Tsuru, *ACS Appl. Mater. Interfaces*, 2017, 9, 8433-8436.
- 171 J. H. Park, M. J. Kwak, C. Hwang, K. N. Kang, N. Liu, J. H. Jang and B. A. Grzybowski, *Adv. Mater.*, 2021, 33, 2101726.
- 172 A. Demessence, P. Horcajada, C. Serre, C. Boissière, D. Grosso, C. Sanchez and G. Férey, *Chem. Commun.*, 2009, 7149.
- 173 I. Stassen, D. De Vos and R. Ameloot, *Chem. Eur. J.*, 2016, 22, 14452–14460.
- 174 L. Sun, G. Yuan, L. Gao, J. Yang, M. Chhowalla, M. H. Gharahcheshmeh, K. K. Gleason, Y. S. Choi,

B. H. Hong and Z. Liu, *Nat. Rev. Methods Primers*, 2021, 1, 5.

- 175 S. Han, R. A. Ciufo, M. L. Meyerson, B. K. Keitz and C. Buddie Mullins, *J. Mater. Chem. A*, 2019, *7*, 19396–19406.
- 176 D. Fischer, A. von Mankowski, A. Ranft, S. K. Vasa, R. Linser, J. Mannhart and B. V. Lotsch, *Chem. Mater.*, 2017, **29**, 5148–5155.
- 177 J. E. Mondloch, W. Bury, D. Fairen-Jimenez, S. Kwon, E. J. DeMarco, M. H. Weston, A. A. Sarjeant, S. T. Nguyen, P. C. Stair, R. Q. Snurr, O. K. Farha and J. T. Hupp, *J. Am. Chem. Soc.*, 2013, **135**, 10294–10297.
- 178 I. Stassen, M. Styles, G. Grenci, H. Van Gorp, W. Vanderlinden,S. De Feyter, P. Falcaro, D. De Vos, P. Vereecken andR. Ameloot, *Nat. Mater.*, 2016, 15, 304–310.
- 179 M. Liu, Y. Liu, J. Dong, Y. Bai, W. Gao, S. Shang, X. Wang, J. Kuang, C. Du, Y. Zou, J. Chen and Y. Liu, *Nat. Commun.*, 2022, **13**, 1411.
- 180 J. Troyano, C. Çamur, L. Garzón-Tovar, A. Carné-Sánchez, I. Imaz, D. Maspoch and D. Maspoch, Acc. Chem. Res., 2020, 53, 1206–1217.
- 181 A. Carné-Sánchez, I. Imaz, M. Cano-Sarabia and D. Maspoch, *Nat. Chem.*, 2013, 5, 203–211.
- 182 Y. Mitsuka, K. Nagashima, H. Kobayashi and H. Kitagawa, *Chem. Lett.*, 2016, **45**, 1313–1315.
- 183 L. Garzón-Tovar, M. Cano-Sarabia, A. Carné-Sánchez, C. Carbonell, I. Imaz and D. Maspoch, *React. Chem. Eng.*, 2016, 1, 533–539.
- 184 C. Avci-Camur, J. Troyano, J. Pérez-Carvajal, A. Legrand, D. Farrusseng, I. Imaz and D. Maspoch, *Green Chem.*, 2018, 20, 873–878.
- 185 C. Avci-Camur, J. Perez-Carvajal, I. Imaz and D. Maspoch, *ACS Sustainable Chem. Eng.*, 2018, **6**, 14554–14560.
- 186 G. Boix, J. Troyano, L. Garzón-Tovar, C. Camur, N. Bermejo, A. Yazdi, J. Piella, N. G. Bastus, V. F. Puntes, I. Imaz and D. Maspoch, *ACS Appl. Mater. Interfaces*, 2020, 12, 10554–10562.
- 187 G. Boix, X. Han, I. Imaz and D. Maspoch, ACS Appl. Mater. Interfaces, 2021, 13, 17835–17843.
- 188 L. Garzón-Tovar, C. Avci-Camur, D. Rodríguez-San-Miguel, I. Imaz, F. Zamora and D. Maspoch, *Chem. Commun.*, 2017, 53, 11372–11375.
- 189 M. I. Nandasiri, S. R. Jambovane, B. P. McGrail, H. T. Schaef and S. K. Nune, *Coord. Chem. Rev.*, 2016, 311, 38–52.
- 190 B. Valizadeh, T. N. Nguyen and K. C. Stylianou, *Polyhedron*, 2018, **145**, 1–15.
- 191 J. Dhainaut, C. Avci-Camur, J. Troyano, A. Legrand, J. Canivet, I. Imaz, D. Maspoch, H. Reinsch and D. Farrusseng, *CrystEngComm*, 2017, 19, 4211–4218.
- 192 O. Ardelean, G. Blanita, G. Borodi, M. D. Lazar, I. Misan, I. Coldea and D. Lupu, *Int. J. Hydrogen Energy*, 2013, 38, 7046–7055.
- 193 H. Oh, D. Lupu, G. Blanita and M. Hirscher, RSC Adv., 2014, 4, 2648–2651.
- 194 G. Blanita, I. Coldea, I. Misan and D. Lupu, *Int. J. Hydrogen Energy*, 2014, **39**, 17040–17046.

- 195 A. Permyakova, O. Skrylnyk, E. Courbon, M. Affram, S. Wang, U. Lee, A. H. Valekar, F. Nouar, G. Mouchaham, T. Devic, G. De Weireld, J. Chang, N. Steunou, M. Frère and C. Serre, *ChemSusChem*, 2017, **10**, 1419–1426.
- 196 P.-J. Kim, Y.-W. You, H. Park, J.-S. Chang, Y.-S. Bae, C.-H. Lee and J.-K. Suh, *Chem. Eng. J.*, 2015, **262**, 683–690.
- 197 V. F. D. Martins, R. Seabra, P. Silva, A. M. Ribeiro, K. H. Cho, U.-H. Lee, J.-S. Chang, J. M. Loureiro, A. E. Rodrigues and A. Ferreira, *Ind. Eng. Chem. Res.*, 2020, 59, 10568–10582.
- 198 R. Girimonte, B. Formisani and F. Testa, *Powder Technol.*, 2017, **311**, 9–17.
- 199 D. E. Crawford, Beilstein J. Org. Chem., 2017, 13, 65-75.
- 200 X. M. Liu, L. H. Xie and Y. Wu, *Inorg. Chem. Front.*, 2020, 7, 2840–2866.
- 201 X. Guo, L. Qiao, S. Zong, R. Ye, Y. He, J. Cheng, X. Cao, Z. Zhou and Y. Yao, *ChemistrySelect*, 2022, 7, e202201574.
- 202 L. E. Burris and M. C. G. Juenger, *Constr. Build. Mater.*, 2020, **258**, 119988.
- 203 E. Hastürk, S. P. Höfert, B. Topalli, C. Schlüsener and C. Janiak, *Microporous Mesoporous Mater.*, 2020, **295**, 109907.
- 204 P. Küsgens, A. Zgaverdea, H. Fritz, S. Siegle and S. Kaskel, J. Am. Ceram. Soc., 2010, 93, 2476–2479.
- 205 Y. Khabzina, J. Dhainaut, M. Ahlhelm, H.-J. Richter, H. Reinsch, N. Stock and D. Farrusseng, *Ind. Eng. Chem. Res.*, 2018, 57, 8200–8208.
- 206 W. Y. Hong, S. P. Perera and A. D. Burrows, *Microporous Mesoporous Mater.*, 2015, 214, 149–155.
- 207 S. Muley, T. Nandgude and S. Poddar, *Asian J. Pharm. Sci.*, 2016, **11**, 684–699.
- 208 R. Gandhi, C. Lal Kaul and R. Panchagnula, *Pharm. Sci. Technol. Today*, 1999, 2, 160–170.
- 209 A. Abramova, N. Couzon, M. Leloire, P. Nerisson, L. Cantrel, S. Royer, T. Loiseau, C. Volkringer and J. Dhainaut, ACS Appl. Mater. Interfaces, 2022, 14, 10669–10680.
- 210 H. Molavi, K. Mirzaei, M. Barjasteh, S. Y. Rahnamaee, S. Saeedi, A. Hassanpouryouzband and M. Rezakazemi, *Nano-Micro Lett.*, 2024, 16, 272.
- 211 C.-T. Hsieh, K. Ariga, L. K. Shrestha and S. Hsu, *Biomacromolecules*, 2021, 22, 1053–1064.
- 212 G. J. H. Lim, Y. Wu, B. B. Shah, J. J. Koh, C. K. Liu, D. Zhao, A. K. Cheetham, J. Wang and J. Ding, *ACS Mater. Lett.*, 2019, **1**, 147–153.
- 213 A. K. Hołda and I. F. J. Vankelecom, J. Appl. Polym. Sci., 2015, 132.
- 214 S. L. Duraikkannu, R. Castro-Muñoz and A. Figoli, *Colloid Interface Sci. Commun.*, 2021, **40**, 100329.
- 215 G. W. Peterson, J. J. Mahle, T. M. Tovar and T. H. Epps, *Adv. Funct. Mater.*, 2020, **30**, 2005517.
- 216 J. Wang, J. Yang, H. Zhu, B.-G. Li and S. Zhu, *Chem. Eng. J.*, 2023, **456**, 141026.
- 217 K. Kiaei, M. T. Nord, N.-C. Chiu and K. C. Stylianou, *ACS Appl. Mater. Interfaces*, 2022, **14**, 19747–19755.
- 218 B. Valizadeh, T. N. Nguyen, S. Kampouri, D. T. Sun, M. D. Mensi, K. Stylianou, B. Smit and W. L. Queen, J. Mater. Chem. A, 2020, 8, 9629–9637.

- 219 S. Yang, L. Peng, O. A. Syzgantseva, O. Trukhina, I. Kochetygov, A. Justin, D. T. Sun, H. Abedini, M. A. Syzgantseva, E. Oveisi, G. Lu and W. L. Queen, *J. Am. Chem. Soc.*, 2020, 142, 13415–13425.
- 220 Y. Wu, Y. Li, Y. Zheng, J. Yao and J. Gao, *Cellulose*, 2023, **30**, 10427–10436.
- 221 T. D. Bennett, A. L. Goodwin, M. T. Dove, D. A. Keen, M. G. Tucker, E. R. Barney, A. K. Soper, E. G. Bithell, J.-C. Tan and A. K. Cheetham, *Phys. Rev. Lett.*, 2010, **104**, 115503.
- W. Xu, N. Hanikel, K. A. Lomachenko, C. Atzori, A. Lund,
 H. Lyu, Z. Zhou, C. A. Angell and O. M. Yaghi, *Angew. Chem.*, 2023, 135, e202300003.
- L. Peng, S. Yang, S. Jawahery, S. M. Moosavi, A. J. Huckaba,
 M. Asgari, E. Oveisi, M. K. Nazeeruddin, B. Smit and
 W. L. Queen, *J. Am. Chem. Soc.*, 2019, 141, 12397–12405.
- 224 H. Yin, H. Kim, J. Choi and A. C. K. Yip, *Chem. Eng. J.*, 2015, 278, 293–300.
- A. J. Howarth, Y. Liu, P. Li, Z. Li, T. C. Wang, J. T. Hupp and
 O. K. Farha, *Nat. Rev. Mater.*, 2016, 1, 1–15.
- 226 S. Karak, K. Dey and R. Banerjee, *Adv. Mater.*, 2022, 34, 2202751.
- 227 Z. Liu, H. Wang, J. Ou, L. Chen and M. Ye, *J. Hazard. Mater.*, 2018, **355**, 145–153.
- 228 Y. Zhu, L. Chen, J. Pan, S. Jiang, J. Wang, G. Zhang and K. Zhang, *Prog. Mater. Sci.*, 2025, 148, 101373.
- 229 X. Chen, D. Menon, X. Wang, M. He, M. R. A. Kiapi, M. Asgari, Y. Lyu, X. Tang, L. L. Keenan, W. Shepard, L. H. Wee, S. Yang, O. K. Farha and D. Fairen-Jimenez, *Chem*, 2025, 102382.
- 230 N. C. Burtch, H. Jasuja and K. S. Walton, *Chem. Rev.*, 2014, 114, 10575–10612.

- 231 H. Yang, H. Wu, F. Pan, Z. Li, H. Ding, G. Liu, Z. Jiang,
 P. Zhang, X. Cao and B. Wang, *J. Membr. Sci.*, 2016, 520, 583–595.
- 232 B. Wang, L. Shen, Y. He, C. Chen, Z. Yang, L. Fei, J. Xu,
 B. Li and H. Lin, *Small*, 2024, 20, 2310174.
- 233 S. Chakraborty, D. Menon, I. Mikulska, C. Pfrang, D. Fairen-Jimenez, S. K. Misra and I. Lynch, *Nat. Rev. Mater.*, 2025, 10, 167–169.
- 234 S. Yang, V. V. Karve, A. Justin, I. Kochetygov, J. Espín, M. Asgari, O. Trukhina, D. T. Sun, L. Peng and W. L. Queen, *Coord. Chem. Rev.*, 2021, 427, 213525.
- 235 G. Li, S. Zhao, Y. Zhang and Z. Tang, Adv. Mater., 2018, 30, 1800702.
- 236 L. Guo, Z. Shen, C. Ma, C. Ma, J. Wang and T. Yuan, *J. Alloys Compd.*, 2022, **906**, 164375.
- 237 P. Bhadane, P. Mahato, D. Menon, B. K. Satpathy, L. Wu, S. Chakraborty, P. Goyal, I. Lynch and S. K. Misra, *Environ. Sci.: Nano*, 2024, 11, 2385–2396.
- 238 D. Ursueguía, E. Díaz and S. Ordóñez, *Microporous Mesoporous Mater.*, 2023, **360**, 112712.
- 239 Y. Li, G. Wen, J. Li, Q. Li, H. Zhang, B. Tao and J. Zhang, *Chem. Commun.*, 2022, **58**, 11488–11506.
- 240 S. Li, W. Han, Q. F. An, K. T. Yong and M. J. Yin, *Adv. Funct. Mater.*, 2023, **33**, 2303447.
- 241 C. Zhou, H. Li, H. Qin, B. Yuan, M. Zhang, L. Wang,
 B. Yang, C. Tao and S. Zhang, *Chem. Eng. J.*, 2023, 467, 143394.
- 242 S. Zhuang, H. Huang, Y. Xiao, Z. Zhang, J. Tang, B. C. Gates and D. Yang, *J. Catal.*, 2021, **404**, 128–138.
- 243 L. S. Xie, G. Skorupskii and M. Dincă, *Chem. Rev.*, 2020, 120, 8536–8580.